

# **Ultra Low Emission Zone**

**Integrated Impact Assessment** 

October 2014



## **Document control sheet**

BPP 04 F8 version 10 October 2014

Project:	Ultra Low Emission Zone Integrated Impact Assessment			
Client:	Transport for London	Project Number:	B1993000	
Document Title:	Integrated Impact Assessment			
Ref. No:	N/A			

		Origi	nated by	Checked by	Reviewed by	
ORIGINAL		NAME		NAME	NAME	
		Michelle Cottrell/ Mason Baker		John Pounder	Peter Marsder	Peter Marsden
Appro	Approved by			As Project Manager I confi above document(s) have b		2
		Rose	lle Bellion	Jacobs' Check and Review that I approve them for is	v procedure and	$\sum$
DATE 24/10/14 Document sta		tus: Final				

REVISION	NAME	NAME	NAME
Approved by	NAME	As Project Manager I confirm that t above document(s) have been sub	he INITIALS
		Jacobs' Check and Review proced that I approve them for issue	
DATE	Docum	ent status:	

REVISION	NAME	NAME NAME	
Approved by	NAME	As Project Manager I confirm that the above document(s) have been subjected to	INITIALS
		Jacobs' Check and Review procedure and that I approve them for issue	
DATE Documen		Document status	

#### Jacobs U.K. Limited

This document has been prepared by a division, subsidiary or affiliate of Jacobs U.K. Limited ("Jacobs") in its professional capacity as consultants in accordance with the terms and conditions of Jacobs' contract with the commissioning party (the "Client"). Regard should be had to those terms and conditions when considering and/or placing any reliance on this document. No part of this document may be copied or reproduced by any means without prior written permission from Jacobs. If you have received this document in error, please destroy all copies in your possession or control and notify Jacobs.

Any advice, opinions, or recommendations within this document (a) should be read and relied upon only in the context of the document as a whole; (b) do not, in any way, purport to include any manner of legal advice or opinion; (c) are based upon the information made available to Jacobs at the date of this document and on current UK standards, codes, technology and construction practices as at the date of this document. It should be noted and it is expressly stated that no independent verification of any of the documents or information supplied to Jacobs has been made. No liability is accepted by Jacobs for any use of this document, other than for the purposes for which it was originally prepared and provided. Following final delivery of this document to the Client, Jacobs will have no further obligations or duty to advise the Client on any matters, including development affecting the information or advice provided in this document.

This document has been prepared for the exclusive use of the Client and unless otherwise agreed in writing by Jacobs, no other party may use, make use of or rely on the contents of this document. Should the Client wish to release this document to a third party, Jacobs may, at its discretion, agree to such release provided that (a) Jacobs' written agreement is obtained prior to such release; and (b) by release of the document to the third party, that third party does not acquire any rights, contractual or otherwise, whatsoever against Jacobs and Jacobs, accordingly, assume no duties, liabilities or obligations to that third party; and (c) Jacobs accepts no responsibility for any loss or damage incurred by the Client or for any conflict of Jacobs' interests arising out of the Client's release of this document to the third party.



1	Intro	duction
	1.1.1	Transport for London (TfL) commissioned Jacobs to undertake a number of assessments, including environmental, health impact, equality impact and economic and business impact, to identify and assess impacts of the proposed Ultra Low Emission Zone (ULEZ).
	1.1.2	The ULEZ is a proposal to reduce emissions specifically from road transport. The following objectives for ULEZ were proposed in line with the characteristics set out in the Mayor's Transport Strategy (MTS):
		<ul> <li>reduce air pollutant emissions from road transport, particularly those with greatest health impacts, to support Mayoral strategies and contribute to achieving compliance with European Union (EU) limit values (LV);</li> <li>reduce carbon dioxide (CO<sub>2</sub>) emissions from road transport, to support Mayoral strategies and contribute to a London-wide reduction; and</li> <li>stimulate the low emission vehicle market by increasing the proportion of low emission vehicles and promoting sustainable travel.</li> </ul>
	1.1.3	<ul> <li>The proposal comprises the following key components:</li> <li>TfL buses: a requirement that all taxis and new private hire vehicles (PHVs) presented for licensing from 2018 would need to be zero emission capable;</li> <li>taxis and PHVs: a requirement that all taxis and new PHVs presented for licensing from 1 January 2018 would need to be zero emission capable alongside an accompanying reduction in the age limit for all non zero emission capable taxis from 7 September 2020 from 15 to 10 years (irrespective of date of licensing); and</li> <li>emission standards for vehicles (the ULEZ standards): to encourage the uptake of cleaner vehicles. From 7 September 2020 vehicles that do not meet ULEZ emissions standards would be required to pay a daily ULEZ charge to drive within the ULEZ area</li> </ul>
	1.2	Purpose of the Integrated Impact Assessment (IIA)
	1.2.1	The Integrated Impact Assessment (IIA) considers and documents the findings of the following individual technical assessments to provide a streamlined and integrated overview of the anticipated effects of the proposed ULEZ:
		<ul> <li>Environmental Assessment (EA);</li> <li>Health Impact Assessment (HIA);</li> <li>Equality Impact Assessment (EqIA); and</li> <li>Economic and Business Impact Assessment (EBIA).</li> </ul>



1.2.2 Reference should be made to each technical assessment report for further detail on any of the individual topics.

## 1.3 Requirement for IIA

- 1.3.1 The proposed ULEZ would be implemented through existing policies covered in *The London Plan* 2011 (including revised early minor alterations to the London Plan, October 2013) and the MTS, which have both undergone Strategic Environmental Assessment (SEA).
- 1.3.2 Nonetheless, TfL's view is that the implications of ULEZ would be best understood through full assessments of the ULEZ in relation to the environment, health, equality and economic and business. The IIA brings together the findings of each of these assessments into one concise integrated document.
- 1.3.3 TfL considers that undertaking an IIA is critical to clearly articulate and identify key impacts associated with the proposed ULEZ including how identified negative impacts can be avoided, remedied or mitigated where possible, and positive impacts enhanced.

## 1.4 Structure of this report

- 1.4.1 The report is structured as follows:
  - Chapter 2 provides background information on the ULEZ;
  - Chapter 3 provides information on the ULEZ and the ULEZ study area;
  - Chapter 4 provides context on the approach of the IIA;
  - Chapter 5 provides a summary of the assessment outcomes of the ULEZ;
  - Chapter 6 provides a summary of the assessment against each IIA objective and whether it has been / will be met by the proposed ULEZ;
  - Chapter 7 presents the conclusions and further recommendations;
  - Chapter 8 provides references; and
  - Chapter 9 provides confirmation of the acronyms used throughout this report.



## 2 Background

#### 2.1 Overview

- 2.1.1 In February 2013, the Mayor requested that TfL examined the feasibility of, and prepared proposals for, the introduction of the ULEZ in central London (Greater London Authority, 2013). The aim of this initiative was to contribute towards the meeting of existing CO<sub>2</sub> targets, nitrogen dioxide (NO<sub>2</sub>) and particulate matter (PM<sub>10</sub>) limit values, and to reduce the high contribution to emissions and concentrations that emanate from road transport. As stated in the Greater London Authority's (GLA's) *2020 Vision: The Greatest City on Earth* (2013), the ULEZ would aim to 'restrict central London only to those vehicles that have zero or near-zero tailpipe emissions'.
- 2.1.2 Whilst the London Low Emission Zone (LEZ), introduced in 2008, and other Mayoral policies have improved air quality in Greater London, the challenge remains to meet the specified air quality limits set by the EU. Air pollution affects the quality of life of a large number of Londoners, especially those with respiratory and cardiovascular conditions. In 2008, an equivalent of 4,300 deaths in the Capital were attributed to long-term exposure to fine particulate matter (PM<sub>2.5</sub>) and a permanent reduction of 1µg/m<sup>3</sup> would increase life expectancy across the population, with the expected gains differing by age (GLA, 2010).
- 2.1.3 A number of strategies published by the Mayor of London, including the Mayor's Air Quality Strategy 2010 (MAQS) and the MTS, aim to reduce emissions to mitigate climate change and improve London's air quality. Since the publication of the MTS, TfL has delivered a greener bus fleet, encouraged the use of electric cars and increased public transport patronage, alongside cycling and walking.
- 2.1.4 TfL's *Transport Emissions Roadmap* 2014 (TERM) builds on preceding strategies and policies by focussing on reducing emissions from ground-based transport in London. The TERM introduces a range of proposed measures to be considered by Government, GLA, TfL and London boroughs to help meet the challenge of reducing CO<sub>2</sub> emissions and air pollutants, particularly oxides of nitrogen (NO<sub>x</sub>), NO<sub>2</sub> and PM<sub>10</sub>, in London.
- 2.1.5 Implementation of the ULEZ in central London is one of the measures identified by TERM.
- 2.1.6 The objectives of the ULEZ are to:
  - reduce air pollutant emissions from road transport, particularly those with greatest health impacts, to support Mayoral Strategies and contribute to achieving compliance with EU limit values;
  - reduce CO<sub>2</sub> emissions from road transport to support Mayoral strategies and contribute to London-wide reduction; and
  - promote sustainable travel and stimulate the low emission vehicle economy by increasing the proportion of low emission vehicles in London.



- 2.1.7 It is proposed that the ULEZ would take effect from 7 September 2020 and operate for 24 hours a day, seven days a week. The geographical scope of the ULEZ would be enforced within the limits of the current Congestion Charge Zone (CCZ), which covers the City of London in its entirety (aside from a small area near to Tower Hill), and partially covers to varying extents the City of Westminster and the London boroughs of Camden, Hackney, Islington, Lambeth, Southwark and Tower Hamlets. This area also experiences the highest levels and concentrations of pollution within London to which the greatest number of people are exposed.
- 2.1.8 The CCZ provides an existing boundary for central London, shaped by the Inner Ring Road (IRR) and well embedded in road user travel behaviour. Not only is this zone a defined area, TfL already operates a comprehensive CCZ camera enforcement network that is also planned to be utilised to manage compliance with the ULEZ.

## 2.2 Development of the ULEZ

2.2.1 The ULEZ has been developed in a way which seeks to avoid or minimise adverse impacts on the environment, health and wellbeing, population and equality groups and London's economy and Small to Medium Sized Enterprises (SMEs). Further information on the development of ULEZ including options appraisal and associated feasibility work undertaken can be found in the ULEZ Supplementary Information Report published by TfL alongside this IIA.



## 3 Details of the Proposed ULEZ

- 3.1.1 ULEZ would set an emissions requirement for all types of vehicles entering central London with charges for non-compliance, discouraging all but the cleanest vehicles. The proposed ULEZ covers three vehicle categories:
  - procurement requirements on TfL buses;
  - revised licensing requirements for taxis (black cabs) and PHVs; and
  - an emissions-based vehicle charging scheme with charges for noncompliance discouraging all but the cleanest vehicles.
- 3.1.2 The proposed ULEZ requirement by vehicle type can be found in Table 3-A and a breakdown of the ULEZ emission standard for each type of vehicle is provided in Table 3-B.

Category	Vehicle	Proposed ULEZ requirement
TfL buses entering ULEZ	TfL double-decker buses	Euro VI hybrid
	TfL single-decker buses	Zero emission at source
Revised licensing	Taxis	10 year taxi age limit
London wide		All newly licensed taxis to be zero emissions capable from 2018
	PHVs	<ul> <li>All newly manufactured/ newly licensed PHVs to be zero emissions capable from 2018</li> <li>All newly licensed second hand PHVs must meet the ULEZ standards</li> <li>Existing licensed PHVs that do not meet the ULEZ standards</li> </ul>
		ULEZ standards must pay the charge when driving in the ULEZ.
Emission-based vehicle charging in	Heavy goods vehicles (HGVs)	Euro VI engine (or pay charge when driving in the ULEZ)
ULEZ	Non-TfL buses and coaches	
	Light goods vehicles (LGVs)	• Euro 4 engine (petrol) or Euro 6 engine (diesel) (or pay charge when driving in the
	Cars and PHVs	ULEZ)
	Motorcycles and power two-wheelers	Euro 3 engine (or pay charge when driving in the ULEZ)

Table 3-A ULEZ proposals by vehicle type



Vehicle type	Proposed emissions standard <sup>1</sup>	Date from when manufacturers must sell or have sold new vehicles meeting the emissions standards	Maximum age of vehicle by 2020 <sup>2</sup>	Charge if vehicle is not compliant with the ULEZ standard <sup>3</sup>
Motorcycle, moped etc.	Euro 3	From 1 July 2007	13 years	£12.50
Car, PHV and small van	Euro 4 (petrol)	From 1 January 2006	14 years	£12.50
	Euro 6 (diesel)	From 1 September 2015	5 years	
Large van and minibus	Euro 4 (petrol)	From 1 January 2007	13 years	£12.50
	Euro 6 (diesel)	From 1 September 2016	4 years	
HGV	Euro VI	From 1 January 2014	6 years	£100
Bus/coach	Euro VI	From 1 January 2014	6 years	£100

Euro standards for heavy-duty diesel engines use Roman numerals and for light-duty vehicle standards use Arabic numerals.

<sup>2</sup> Vehicles this age or younger in 2020 will comply with the ULEZ standard and not incur a charge. <sup>3</sup> This is poughts in addition to any applicable LEZ or COZ observes.

This is payable in addition to any applicable LEZ or CCZ charges.

#### Table 3-BULEZ standard for each type of vehicle

- 3.1.3 It is proposed that a small number of vehicle types would be exempt from the ULEZ charge, including agricultural, military and historic vehicles<sup>1</sup> and non-road going vehicles which are allowed to drive on the highway. These vehicles typically use engines certified to different standards than road-going engines and/or are unsuitable for conversion to an alternate fuel or engine replacement. Further improvements to these vehicles, particularly non-road going vehicles, are pursued through other strategies, such as the MAQS.
- 3.1.4 Taxis (black cabs) that are licensed with London Taxi and Private Hire will also be exempt from the ULEZ charge. Instead, TfL proposes to reduce the age limit of all non zero emission capable taxis from 15 years to 10 years in 2020 to help reduce NO<sub>x</sub> emissions from the taxi fleet and encourage the uptake of zero emission capable taxis (which would retain a 15 year age limit).
- 3.1.5 TfL has committed to the following mitigation measures for the proposed ULEZ:
  - For residents in the ULEZ all residents living in the ULEZ area will be granted a three year time limited 100% discount to recognise that they are unable to avoid the ULEZ area and may require more time to change their vehicle for one to meet ULEZ emissions standards.
  - For taxis should a reduction in the taxi age limit be taken forward as a result of ULEZ TfL will establish a specific fund for drivers of older taxis to help them switch to newer vehicles. It is anticipated that grants would be offered to eligible taxi owners and that the proposed scheme would be phased from 2018 to smooth the impact of a reduced age limit in 2020.

<sup>&</sup>lt;sup>1</sup> Vehicles manufactured before 1973.



3.1.6 These mitigation measures have been treated as an integral part of the ULEZ for the purposes of our assessment of the potential impacts of the proposal on the environment, health and wellbeing, population and equality groups and London's economy and SMEs.

### 3.1.7 The ULEZ study area

- 3.1.8 The study area for the ULEZ falls within the Greater London Administrative Area (GLAA). In some instances, areas beyond the GLAA were considered, as changes to vehicle trip patterns on London's road network brought about by implementation of the ULEZ are likely to extend beyond this boundary.
- 3.1.9 The study area is divided into five zones as described in Table 3-C, which correspond to those employed in the atmospheric emissions modelling that informed the development of the ULEZ.

Zone	Extent
CCZ	Based on the existing boundary which has been in operation since 2003
Inner Ring Road (IRR)	A 12 mile (19km) route formed from a number of major roads that encircle the CCZ
Inner Zone	Extends outwards from the CCZ to cover a number of London boroughs including Haringey to the north, Newham to the east, Lambeth to the south and Hammersmith and Fulham to the west
Outer Zone	Extending from the boundary of the Inner Zone to the boundary of the GLAA. Includes London boroughs such as Enfield to the north, Havering to the east, Croydon to the south and Hillingdon to the west
Non-GLAA	Covers the area outside the GLAA boundary

 Table 3-C
 Description of the five zones making up the ULEZ study area

- 3.1.10 The same study area was adopted across all assessment reports including the EA, HIA, EqIA and EBIA.
- 3.1.11 A map of the CCZ, inner, outer and non-GLAA zones covering the IIA study area is provided in Figure 3-A. Although not shown on the map, the IRR sits on the boundary between the CCZ and Inner Zone.
- 3.1.12 With the exception of the IRR (which defines the boundary of ULEZ), the four zones are consistent with the London Atmospheric Emissions Inventory (LAEI) 2010.

HEMEL Potters Chinard		ham Green	Wormley 1		Bobbingworth Bike
Felden HEMPSTEAD Crouch Green		128 West End	id Holyfield Bumble's	-Epping Common	North Weald Bassett High Ongar
lovingdon Kings Langley	Frogmore 125 Chiney 5 M South Minms 5	Northaw 177 1250 711	CHESHUNT - Hayes Hill Farm	Upland EPPING Fiddlers Toot	Hill Greensted ONGAR Nine Blackmon
Chipperfield Commonwood Control Abbbits Bricket	RADLETT Shenley RADLETT Shenley Ridge	POTTERS C Cuffley	Waltham Upshi	le 4 Hamiet	A113 Ashes Stondon Begg
Beisize Bucks:Hill Sarrätt Chandler's	denham Green Vroti	A 1005 Botany Bay			Stapleford Kelvedon Doddinghur
Chenies Cross A411	ASTRIA Price	1471 Enfield Chase Enfield	Sewardstone High Beach	Bois A 13	Hatch 104 Navestock Navestock Mountnes
CHORLEY-WOOD Green Buskey		Hadley ENFIELD Cockfosters	Ponders Pinders	LOUGHTON Abridge Bridge	Lableford Green
RICKMANSWORTH	Bushey Add 1 Arkley BARNI	LEAST CONTRACTOR	Blekhurs		Athotts Pilgrims Hatch
Heronsgate	Evoton Carleway Services Mill Hill	FRIERN BARNET	CHINGFORD Hill	Grange Hill Chigwen Bow	Hill Harold
haple cross / NORTH WOOD Haid	A41		Grden W WOODFORD W	Painault A 111	atte-Boyer Hill II Brock Street Warley
Chalfont St.Peter A41 A41 A41 A41 A41 A41 A41 A41 A41 A41	Wealdstone HENDO		WALTHAMSTOW	arkingside	Gidea Harold P25 Little Warley Wood G Childen
M25 Denham Common Green Free Free Free Free Free Free Free		Golders HORNSEY	A 04	Seven ROMFORD	HORNCHURCH
Denham	Harrow Classical Control of the Still	HAMPSTEAD NEWINGTON	LEYTON CHANGTEAD		RISTY A124
imer MO Charles Charle	WEMBLEY Station	CAMPEN ISLINGTON		BARKING	Park 17
Heath UxBRIDGE	Perivale	FINSBURY -	BETHNAL STRATFORD	ST CHARLES A	South South
Cowley Verding Cherry Add 20		A SHOR	EDITCH BOW, MARKEN HAM		Cocken
Grand Charles HAYES SOUTHALL	ACTON ACTON ACTON ACTON ACTON ACTON	10 Nrde Parx	T BOPLAB		HAINHAW Wernington
Argiev Park WEST Norwood Green	BRENTFORD HAMMERSMITH	KENSINGTON TAMBETH	AUNUSEY Dogs Charlt	MATH WOOLWICH	dere Purfleet
Harmondsworth Cranford Services Pa		M BATTERSEA		Plunistead S R	TH Stady Pore Contract GR
Poyle	ISLEWORTH SOL		DERTFORD Baskheath, P Kidbrodke	ABILITHAT	GrayFORD Greenhithe SWANS
Wraysbury Stanwell 14 Noor Stewell East Hythe Bedfor	RICHMOND PUTNEY	WANDSWORTH	LEWISHAM Hither 3	ELTHAM BEX EY Red Ha	DARITORI Bluewater Retail Centre o
ELTHAM	TWICKENHAM Bark	Upper STREATHAM	A Sydenham		thlow Bean A2
IAM FLASHEORD Hanworth	TEDDINGTOWN	Norwood Crystal	The second second	Foots Cray Cray	Wilfmington Lane End Betsham Havitey Darenth
Thorpe Laloham	Clushy Park		BROMLEY	St Paul's	Sutton at Hone Longfi
Weltron Waltron Than	Stampton Court MOR	RDEN A MITCHAN - Sou	Eten Park Bickley	Petts	A225 Kirby Hartley
Lyne Shepherton ON- DL	A 09 Long Dition		West Wickfram	armon	Farningham A20 A20 P 2 Ne ROMAN - Castler A20 M20 Fawkham Ash 0
ADDLESTONE AS	ESTER Hook 24 Park-	CARSHALLON BENDINGTON CARSHALLON	I ON A Additional Vestor	Farnborough Chelsfie d	Eunstord Castle
A319 A320 A320 A320 A320 A320 A320 A320 A320	Chessington		A2022 New Addinaton	Green 7	West Hatch Culver
Woodham A245 West Byffeet	PSOM PSOM	PURIEY	Selsdon Leaves	Downe Fratt's Houng	Shoreham
Pyrford Stoke	A244	STEAD	Bigg	Cudham	Otford Difortines Way Wrotnam
Wo KING Priory Alerion Action	5 A24 A24	Burgh Heath	WÀR LÍNGHAM A Tatsne	Horns Chevening	Kemsing M26 0
Westfield	Ashtead Tadworth	Shipstead Shipstead	Woldingham Garden Village Woldingham	Pilorina Contraction	Ightham Borou
Send A247 Burntcommon File East	ham Headley Hill	Chalden Chalden CALERHAT		ckettane 10 Frasted	Riverhead SEVENOAKS Hy Hatch
West Horsley Horsley A246 West Hatchlands B Clandon Park	Great Bookham R Mickleham	M¢rstnam	Limpsfield		hitley A225 June Suber Suber Soar, Mahor
Clandón Park	Westhumble	A242 REDHILL	Dxted	urt 235 de 44-e	Shipba 0 0.75 1.5
Activities of a register of the second secon	A25 425		Hurst	Chartwell Toy's Hill	evenoaks Miles

	Figure 3-A	
Loves Green		_
OFP		
gar Hill Mill Gree Fryerning	· Colom	
rst 13	Alle -	
Hey		
A1296	and the second sec	
field Hutton		
A128	a started water	
Herongate	Кеу	_
erditch 3	LAEI 2010 Zones	
West	CCZ Inner London	
Horndon Bulphan	Outer London External (Non-Greater London Area)	
9		
n ndon		
Baker Street		
d Little Thurrock		
S		
RAYS CAL		
ALS A		
Sin Sin		
Southfleet Hook Green		
field Rise New Barn Street		
Meopham	0 23/10/14 Created to aid integrated assessment BW MC RK MB	3
Station Station Green	Rev.         Date         Purpose of revision         Drawn         Checkd         Revid         Appr           JACOBS*	'd
Ridley	Tower Bridge Court, 224/226 Tower Bridge Road London, SE1 2UP,UK. Tel: +44(0)20 7403 3330 Fax:+44(0)20 7939 1418 www.jacobs.com	
Green Vigo	Client Transport for London	-
seat 2017 Trothe	Project	
	Ultra Low Emission Zone - Integrated Impact Assessment	
Batt Heath	Drawing Title The ULEZ IIA Study Area	-
ugh Crouch		
Claygate Cross 168	Drawing Status         FINAL           Scale @A3         1:200,000         DO NOT SCAL	E
Dunk's Green	Jacobs No.         B19930000         Client No.           Filepath         Y:B19930000 UE27MidsFigure3A_JIA.mod*           Drawing No.         Fileware 0.4	_
2.25 3 3.75	Figure 3-A This drawing is not to be used in whole in or part other than for the intended purpose and project as defined on this drawing. Refer to the contract for full terms and conditions.	-
- II 🔪 II 🖛 🚿		



## 4 Approach to the IIA

### 4.1 Overview of the IIA process

4.1.1 The IIA is a means by which the common aspects of different technical assessments are brought together in a holistic and integrated manner. An overview of the focus for the four assessments is provided in Table 4-A.

Assessment	Focus
EA	Identifies and assesses the impacts across a range of environmental issues as a result of the ULEZ including: air quality, noise, climate change, biodiversity and nature conservation, cultural heritage, landscape, townscape and the urban realm, material resources and wastes.
HIA	Identifies and assesses impact on the health and wellbeing of the population of Greater London and the ability to access health-related facilities and services as a result of ULEZ. The assessment also addresses equalities issues and thus has some overlap with the EqIA.
EqIA	Identifies and assesses impacts on equalities issues, in particular those groups of people with protected characteristics <sup>1</sup> or are socio-economically disadvantaged.
EBIA	Identifies and assesses impacts on London's economy as a result of the ULEZ and identifies potential impacts on small to medium sized enterprises (SMEs).

<sup>1</sup> people with protected characteristics are defined in the Equality Act 2010

#### Table 4-A Overview of technical assessments undertaken for the ULEZ

- 4.1.2 The requirement for the HIA is driven by emerging guidance and policy requirements, and elements of the international SEA Protocol which came into effect on 11<sup>th</sup> July 2010. The TfL *Health Action Plan* 2014 requires TfL to investigate health impacts of its work; therefore, in line with precedents set by similar transport schemes, TfL required an HIA to be completed to determine the potential health implications of the proposed ULEZ.
- 4.1.3 As a public body, TfL must meet certain requirements in relation to the *Equality Act 2010* when developing proposals and policy. On this basis, an EqIA has been completed to identify any potential disproportionate or differential impacts from the implementation of the ULEZ on people with protected characteristics as defined in the Act as well as socio-economically deprived communities.
- 4.1.4 An EBIA was considered beneficial by TfL to better understand the potential impacts on London's economy and on SMEs as a result of the ULEZ and to inform the decision making process in the development of the ULEZ.
- 4.1.5 There have been a number of stages involved in completion of the assessments, as shown in Figure 4-A and described in the following sections.





Figure 4-A Stages of the IIA Process



## 4.2 Establishing a framework

- 4.2.1 The IIA framework for the MTS has been employed as a logical starting point for the ULEZ IIA, although appropriate and necessary amendments have also been considered and adopted. The MTS IIA framework consists of the following:
  - Primary Objectives (criteria);
  - Secondary Objectives (sub-criteria); and
  - a set of detailed questions to guide the assessment in applying the criteria (MVA consultancy in association with ERM and Future Inclusion, 2009).
- 4.2.2 Appendix 1 provides the IIA Framework for the ULEZ. This has been employed for the purposes of assessment, and to assist stakeholders in finding information under familiar headings
- 4.2.3 Whilst, the MTS IIA objectives and criteria have been used as a starting point, the following amendments have been made to meet the requirements of ULEZ:
  - objectives reorganised by IIA theme and topic area for greater clarity;
  - questions have been retained which enable the identification of both potential positive and negative effects;
  - methods (e.g. "through", "by", etc.) have been removed from objectives;
  - some topics have been removed from objectives;
  - sub-objectives are used to clarify any complex objectives; and
  - the use of "sustainable" has been removed from objectives, as sustainable development is the overall objective of conducting an assessment such as an IIA (use of phrases such as "sustainable transport" and "sustainable economic progress" within objectives introduces factors across all assessment topics).
- 4.2.4 Each technical assessment addressed one or more IIA topics and assessed the extent to which the ULEZ worked towards achieving the associated objective for the topic.
- 4.2.5 The IIA topics and objectives addressed by each assessment are identified in Table 4-B.
- 4.2.6 A summary of how the proposed ULEZ meets each IIA objective is provided in Chapter 6.



Assessment	IIA Topic	IIA Objective
EA	Air quality	To contribute to a reduction in air pollutant emissions and compliance with EU limit values
	Noise	To reduce disturbance from general traffic noise
	Climate change	To reduce CO <sub>2</sub> emissions and contribute to the mitigation of climate change
	Biodiversity including flora and fauna	To protect and enhance the natural environment, including biodiversity, flora and fauna
	Cultural heritage	To protect and enhance historic, archaeological and socio-cultural environment
	Water	To protect and enhance river spaces and waterways through planning and operation
	Material resources and waste	To promote more sustainable resource use and waste management
	Landscape, townscape and urban realm	To protect and enhance the built environment and streetscape
HIA	Health and wellbeing	To contribute to enhanced health and wellbeing for all within London
EqIA	Population and equality	To enhance equality and social inclusion
EBIA	London's economic competitiveness	Provide an environment which will help to attract and retain internationally mobile businesses
	SMEs	Support the growth and creation of SMEs

#### Table 4-B IIA objectives

### 4.3 Initial scoping and stakeholder consultation

- 4.3.1 Jacobs prepared an initial scoping report to identify the ULEZ study area and inform the technical assessments.
- 4.3.2 Two phases of consultation have been undertaken on the ULEZ:
  - Phase 1 stakeholder workshops on proposed methodology and scoping
  - Phase 2 workshops to review emerging assessment and identify mitigation and enhancement opportunities
- 4.3.3 Phase 1 consultation was undertaken during June and July 2014, which aligned with the scoping phase of the ULEZ and Phase 2 consultation was undertaken in late August and throughout September 2014, which aligned with the assessment phase of the ULEZ.
- 4.3.4 The workshops were held by TfL and included participants from a range of stakeholder groups identified in Table 4-C.
- 4.3.5 Key issues discussed during both phases of consultation are outlined in Table 4-D.
- 4.3.6 Workshop attendees were given the opportunity to provide further written responses to the assessment methodologies and scoping up until the end of Wednesday 2 July 2014.



Group	Stakeholders
Government agencies	<ul> <li>Environment Agency</li> <li>GLA</li> <li>TfL</li> <li>Public Health England</li> </ul>
Local authorities	<ul> <li>City of Westminster</li> <li>City of London</li> <li>London Borough of Hackney</li> <li>London Borough of Islington</li> <li>London Borough of Lambeth</li> <li>London Borough of Westminster</li> <li>London Borough of Southwark</li> <li>London Borough of Camden</li> </ul>
Industry bodies	<ul> <li>Confederation of passenger transport</li> <li>Road Haulage Association</li> <li>Freight Transport Association</li> <li>London Transport Coach Operators</li> <li>Chartered Institute of Logistics and Transport in the UK</li> <li>Cross River Partnership</li> <li>Federation of Small Businesses</li> </ul>
Consultants	<ul><li>Jacobs</li><li>Ben Cave Associates</li></ul>
Interest group	Changing Perspectives

Table 4-C	Stakeholders at workshops
-----------	---------------------------

Assessment	Key issues raised
EA	<ul> <li>More electric vehicles on the market may result in more charging points which may impact the townscape and safety.</li> <li>The effects of 'black carbon' (PM ≤2.5µm) should be included in the environmental assessment (PM<sub>2.5</sub> is already in the scope of HIA).</li> <li>Effects on water resources and water quality should be included within the scope of the EA.</li> <li>Traffic migration/displacement as a result of ULEZ (i.e. vehicles avoiding the zone) will lead to the creation of new noise climates and potential impacts on residents.</li> <li>Adequate road/pavement space may not be available within central London to accommodate the additional cycle parking required following an anticipated increase in the uptake of cycling.</li> <li>The emissions factors to be applied to the concentration modelling were questioned.</li> </ul>
HIA	<ul> <li>Concern around the lack of stakeholder input into the option selection process. The intention of the workshop was to seek stakeholder input into the approach to the IIA/HIA process and not to gauge opinion on the options for the ULEZ.</li> <li>Whether the inter-relationship between health and access to work and education would be covered by the HIA.</li> <li>The opportunity to engage in physical activity should be considered by</li> </ul>
	<ul> <li>the HIA as a measure of health and wellbeing.</li> <li>A pragmatic approach should be considered for the HIA and the scope should be refined as far as possible to maintain and produce a focused</li> </ul>



Assessment	Key issues raised
	assessment.
	<ul> <li>As it is one of the objectives of the ULEZ, health should be at the core of the IIA.</li> </ul>
	<ul> <li>Concerns regarding health service provision and the impacts of ULEZ</li> <li>there are a number of 'just-in-time' deliveries associated with delivering healthcare;</li> </ul>
	<ul> <li>there is a National Health Service (NHS) policy to move towards 'community-based' care (i.e. local) where the staff have rotating shifts which mean they are required to travel between different hospital sites and some hospital sites may span the CCZ;</li> </ul>
	<ul> <li>the concept of social capital was criticised as being dominated be an economic focus. For some people the quality of life is not determined by whether it is possible to get employment. The assessment should ensure that the focus is on having a good quality life.</li> </ul>
EqIA	• Displacement of traffic may be an issue in relation to impacts e.g. air quality surrounding ULEZ boundary, effects on socio-economically deprived areas.
	Perception of community safety in relation to different travel/transport modes.
	<ul> <li>Effect on groups that rely on community vehicles or similar, modes of transport.</li> </ul>
	• Potential impact on availability of wheelchair accessible taxis, PHVs, coaches and minibuses.
	<ul> <li>Impact on availability of taxis and PHVs at night and the impact on safe/use of unlicensed cabs.</li> </ul>
	• Disabled drivers require specific vehicles adjusted to their needs. Wi such compliant vehicles be available to purchase?
	<ul> <li>Black, Asian and Minority Ethnic (BAME) groups represent a high proportion of the SMEs within central London, including market trader within ULEZ and minicab drivers, many using their own vehicle.</li> </ul>
	<ul> <li>Socio-economically deprived groups could be unable to afford compliant vehicles, be unable to pay the charge for a non-compliant vehicle, and be unable to pay the fine for not paying the charge, potentially leading to debt enforcement action. These groups are also potentially more likely to work 'unsociable' hours, when there is more a reliance on travel by car.</li> </ul>
	• Early and proactive communication of the proposed ULEZ to resident is considered to be important.
	• Impacts relating to coach companies transporting pupils to schools within and outside the ULEZ were raised.
	Concern about the cost of mobility adapted vehicles that will need replacing to meet ULEZ standards.
	<ul> <li>Impacts on people travelling into Central London by car or minibus to attend faith services are only likely to be an issue on the edge of ULEZ. Example cited of a large Pentecostal ministry in Elephant and Castle which uses a fleet of minibuses to transport worshippers.</li> </ul>
	<ul> <li>Noted that many religious services are held in non-religious buildings inner and outer London – but not likely to be prevalent in central London.</li> </ul>
	• Are there particular safety issues for the Lesbian, Gay, Bisexual and Transgender (LGBT) community? Are they more likely to use cabs? Highest clusters are Westminster and Lambeth.



Assessment	Key issues raised
EBIA	• The ULEZ is seen as a potential benefit to restaurants and cafes within the ULEZ boundary, as improvements in air quality may encourage customers to use on-street seating.
	• Concerns surrounding the potential wider impact on tourist economy in London and a perception that London is becoming too expensive.
	<ul> <li>Impact will be felt outside central area, as well as in central London. Concerns were expressed about SMEs, in particular, for sectors of the economy which typically have very small fleets, including: breweries/micro-breweries; meat suppliers; farms which deliver produce direct to high-end hotels/restaurants; and, personal services (e.g. plumbers, massage therapists). Margins will be squeezed.</li> </ul>
	<ul> <li>Concern that a loss of small independent businesses would impact on the diversity of London's offer to tourism.</li> </ul>
	• Concerns that there may be impacts on local authority budgets (e.g. contracted out refuse collection vehicles).
	• Concerns that a reduction in black cabs may impact on the night time economy (e.g. through perceptions of personal travel safety), although this may be overcome by technology (e.g. taxi apps).
	• The impact on motorcycle couriers was also raised, although this is unlikely to be an issue due to the present rate of compliance.

Table 4-DSummary of consultation

### 4.4 Assessment

- 4.4.1 The scope and content of each of the assessments were informed by stakeholder workshops.
- 4.4.2 All assessments recommend actions to mitigate possible negative impacts as well as enhancements to maximise positive impacts.

### (a) EA

- 4.4.3 The scope of the EA comprised the assessment of air quality, noise, climate change, biodiversity & nature conservation, cultural heritage, landscape, townscape & urban realm, material resources and wastes.
- 4.4.4 Soil and water were not considered relevant to the EA of the ULEZ and were not included in the scope. Soil can be a sensitive receptor for air pollution deposition. In the context of the urbanised London study area, and given the likely level of air quality changes anticipated, it was deemed that effects to soil would be insignificant. Given the nature of the proposed ULEZ, no measureable effects on the water environment are expected.



## (b) HIA

- 4.4.5 The assessment considered impacts associated with air quality, noise and neighbourhood amenity, accessibility and active travel, crime reduction and community safety, social cohesion and lifetime neighbourhoods, climate change, access to healthcare services and other social infrastructure.
- 4.4.6 The assessment did not include impacts from housing quality and design, access to open space and nature, access to healthy food, access to work and training and minimising the use of resources as none of these would be affected by the proposed ULEZ.

#### (c) EqIA

- 4.4.7 The EqIA assessed the effects of the implementation of the ULEZ on people with protected characteristics as defined in the Equality Act. Specifically the following equality groups were considered in the EqIA: age, disability, sex, race, pregnancy and maternity, gender reassignment, religion or belief, sexual orientation, socio-economically deprived.
- 4.4.8 In relation to the assessment of impacts on equality groups resulting from traffic, noise and air quality, the scope included young and old and socioeconomically deprived as there is potential for differential effects. No differential or disproportionate effect is anticipated on Lesbian, Gay, Bisexual and Transgender (LGBT), sex, disability, religion or belief, pregnancy and maternity and as such they have been excluded from the scope of the assessment.
- 4.4.9 In relation to the assessment of impacts on equality groups resulting from access to transport systems and sustainable modes of travel, all equality groups were included in the scope, however public sector employees (or contractors) who travel during the course of their work to deliver education, social and health care services to people with protected characteristics were not. It was assumed that the cost associated with compliance with the ULEZ will be incurred by the employer.

### (d) EBIA

- 4.4.10 The EBIA was undertaken by vehicle type, examining the potential impacts that the ULEZ may have on vehicle use and then on the relevant sectors of the economy (e.g. retail, construction, catering, markets, care sector, home bases services, night time economy, tourism) as well as on SMEs.
- 4.4.11 The assessment looked at those sectors in the ULEZ study area that have significant dependence on road transport such as construction, retail and the evening economy. It also identified and assessed impacts for those SMEs providing niche services that are dependent on road transport.
- 4.4.12 Most sectors of the economy such as financial and business services which have little dependency on road transport to operate successfully were not assessed. Similarly, SMEs in sectors such as financial and business services that have little dependency on road transport to operate successfully were not assessed.



## 4.5 Reporting

- 4.5.1 The IIA is supported by individual technical reports for each of the four assessments. The technical reports provide greater detail on baseline conditions and the assessment of likely impacts of the ULEZ. Each report contains the following:
  - overview of ULEZ;
  - policy and legislative context;
  - methodology;
  - baseline;
  - assessment;
  - mitigation; and
  - conclusions.

## 4.6 Public consultation

4.6.1 The IIA will be made available as part of the public consultation, which runs from 27<sup>th</sup> October 2014 to 9<sup>th</sup> January 2015.

#### **Assessment Outcomes** 5

#### Key findings 5.1

- 5.1.1 The key findings from each technical assessment including the impacts and mitigation measures are presented in Table 5-A and broken down into IIA topics.
- 5.1.2 For the purposes of the IIA, each impact has been scaled as minor, moderate or major. The legend to assist with interpreting the scale of impacts from ULEZ (whether positive or negative impacts, minor, moderate or major) is provided in Figure 5-A.
- 5.1.3 Each impact has also been assigned a duration which correlate with the timeframes of the proposed ULEZ: short-medium term (being less than 5 years i.e. impact has disappeared by 2025) or long term (being greater than 5 years i.e. impact is still prevalent after 2025).

Legend	
Positive minor	
Positive moderate	
Positive major	
Neutral	
Negative minor	
Negative moderate	
Negative major	

Figure 5-A Legend for summary assessment table

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Air quality (in relation to environmental sensitive receptors <sup>1</sup> )	<ul> <li>The baseline data for the air quality assessment was forecasted pollutant emissions in central London between 2020 and 2025 and showed that without ULEZ:         <ul> <li>In 2020 approximately 34,000 sensitive receptors (including residential properties, care homes, health facilities and schools) across London would be exposed to NO<sub>2</sub> concentrations which exceed the annual mean Air Quality Objective (AQO).</li> <li>In 2020 approximately 34 sensitive receptors in London would be exposed to PM<sub>10</sub> concentrations which exceed AQO.</li> </ul> </li> <li>The baseline forecasts suggest that cars, buses and taxis would be the three main contributors to total road traffic emissions of NO<sub>x</sub>, NO<sub>2</sub> and PM<sub>2.5</sub> in 2020 and 2025. Together these three vehicle types would account for 76% of total NO<sub>2</sub> emissions in central London in both 2020 and 2025.</li> </ul>	<ul> <li>ULEZ would result in improvement in NO<sub>2</sub> concentrations as follows:</li> <li>An improvement in annual average mean concentration of 4.6 µg/m<sup>3</sup> in 2020 and a further 2.3 µg /m<sup>3</sup> in 2025 in the CCZ with a reduction towards the Outer Zone with the improvement potentially lessening to 0.6ug/m<sup>3</sup> in 2020 and a further 0.3 µg /m<sup>3</sup> in 2025.</li> <li>In 2020, 18,000 sensitive receptors across London are forecast to no longer be exposed to concentrations exceeding the annual mean NO<sub>2</sub> AQO, compared to the without ULEZ- a reduction of approximately 52%. A further 2,000 properties would no longer experience exceedances by 2025 (a further 13% reduction).</li> <li>The largest percentage reduction in properties exceeding the AQOs in 2020 would be in central London (approx. 4,500 or 86%), followed by Inner Zone (approx. 10,500 or 52%) and Outer Zone (approx. 2,800 or 33%).</li> <li>Note: air quality impacts relating to other sensitive receptors such as care homes, hospitals and schools is provided in the IIA topic health and wellbeing.</li> </ul>	Positive long term	Major	<ul> <li>The Mayor has outlined a strategy of measures to tackle pollution:         <ul> <li>Making London's bus fleet cleaner.</li> <li>Capping the age of London's taxi and private hire fleet.</li> <li>Setting new and tighter standards for the London LEZ.</li> <li>Encouraging the uptake of electric and other low emission vehicles.</li> <li>Investing record amounts in cycling (TfL, 2014).</li> </ul> </li> <li>The Mayor's 2050 Infrastructure Investment Plan details significant investment in public transport and urban realm.</li> <li>The Mayor is also supportive of:         <ul> <li>The Government to help replace the most polluting diesel cars with lower emission vehicles.</li> <li>A new public awareness programme called Breathe Better Together to tackle air pollution at schools and an ambitious public awareness campaign to be launched late midyear (GLA, 2014).</li> </ul></li></ul>	<ul> <li>TfL should use ULEZ as a benchmark for its wider transport policy-making e.g. the Congestion Charge, their role in planning applications and for TfL fleet.</li> <li>TfL should investigate the potential expansion of ULEZ / raising of ULEZ standards in the future (i.e. post 2025) or the introduction of complementary adjacent zones for low emissions neighbourhoods. This could result in further air quality benefits. Dependent on vehicle market, infrastructure, opportunity and compliance costs at the time of investigation and public consultation.</li> </ul>

averaging period of the objective; for an annual mean objective - this includes receptors where members of the public may be present for 6 months or more.

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Air quality	Refer to text on previous page.	<ul> <li>ULEZ would result in a decrease in PM<sub>10</sub> emissions:</li> <li>Relevant receptor weighted average annual mean PM<sub>10</sub> concentrations reduce by a small amount in the CCZ (0.2µg/m<sup>3</sup>). Moving away from the influence of the ULEZ this reduction is anticipated to be negligible in the Inner Zone and beyond.</li> <li>The City of London would have the largest reduction in relevant receptor weighted average annual mean PM<sub>10</sub> concentration in both 2020 and 2025 – average reduction of 0.1 µg/m<sup>3</sup> per relevant receptor.</li> <li>A single sensitive receptor would change from being exposed to concentrations exceeding the annual mean PM<sub>10</sub> to no longer being exposed to exceedances in 2020. This receptor is in the London Borough of Tower Hamlets.</li> <li>In 2025, no sensitive receptors are anticipated to change from being exposed to concentrations exceeding the AQO to no longer being exposed to exceedances.</li> </ul>	Positive long term	Minor	Refer to text on previous page.	Refer to text on previous page.
		<ul> <li>ULEZ would result in small reductions in PM<sub>2.5</sub> concentrations:</li> <li>PM<sub>2.5</sub> concentrations may reduce by 0.1µg/m<sup>3</sup> in both the CCZ and IRR in 2020. This declines over time, such that by 2025 the reduction in each zone is negligible.</li> <li>No projected change in the number of sensitive receptors exposed to concentrations exceeding the annual mean PM<sub>10</sub> objective without ULEZ baseline for both 2020 and 2025.</li> </ul>	Positive short- medium term	Minor		
Noise	<ul> <li>The main source of ambient noise in London is roads.</li> <li>In urban areas, most vehicle noise comes from engines, because at low speed, engine noise dominates over the noise generated by tyres and road surfaces.</li> <li>41% of Londoners are disturbed by road traffic noise.</li> <li>Studies have found that the use of hybrid buses can</li> </ul>	ULEZ would result in an overall minor reduction in noise levels generated from introduction of zero emission single- decked electric and hydrogen buses and low emission hybrid double-decker buses by around 3dB(A) or 30% compared to conventional diesel based engine buses.	Positive long term	Moderate	<ul> <li>TfL has existing policies to assist with noise reduction from road traffic including policies around smoother driving and no idling for buses.</li> </ul>	<ul> <li>Opportunity for positive publicity campaign to promote public awareness of the benefits of noise for the ULEZ.</li> <li>Engage in early and proactive communication of the ULEZ about noise related safety impacts e.g. any concerns around potential residual impacts of reduced traffic</li> </ul>
	<ul> <li>offer a reduction in noise levels of around 3dB(A) or 30% compared to conventional diesel based engines.</li> <li>Fully electric buses can offer even greater reductions in noise and vibration.</li> </ul>	ULEZ may result in additional noise reductions through the phasing out of older taxis and through mandating all newly licensed taxis and PHVs to be zero emissions capable from 2018.	Positive long term	Minor		noise levels on personal road safety.

IIA topic		Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Climate change	•	<ul> <li>CO<sub>2</sub> is the Greenhouse Gas (GHG) of most relevance to this assessment.</li> <li>Greater London is responsible for approximately 10% of the UK's total CO<sub>2</sub> emissions (2012), of which 17% is attributable to road traffic. Central London accounts for about 1% of Greater London's total CO<sub>2</sub> emission.</li> <li>Without ULEZ London's transport emissions would be expected to reduce to 8.02 million tonnes CO<sub>2</sub> per annum by 2025, a 15% decrease of 1990 levels, despite projected population and employment growth in excess of 10%.</li> <li>ULEZ would result in an increased uptake of low and zero emission vehicles and greater compliance with more stringent EURO fuel standards.</li> </ul>	ULEZ would result in further reductions in total road traffic CO <sub>2</sub> emissions by 123,000 tonnes per annum in 2020 and 169,000 tonnes per annum in 2025 in central London. This equates to overall reductions of 2% in 2020 and 3% in 2025. In central London this will be experienced predominantly within the existing CCZ and to a lesser extent in the IRR and Inner Zone: the local authorities likely to see greatest reductions in CO <sub>2</sub> emissions as a result of the ULEZ are the City of London, City of Westminster and Kensington and Chelsea. Boroughs on the edge of the GLAA boundary would experience smaller reductions.	Positive Long term	Minor	<ul> <li>The Mayor has committed to reducing emissions of CO<sub>2</sub> in London by 60% overall, relative to 1990 levels and across all sectors, by 2025.</li> <li>The Mayor has published strategies on climate change adaptation including the Climate Change Mitigation and Energy Strategy and Climate Change Adaptation Strategy.</li> <li>Other policies relating to climate change include:         <ul> <li>The TERM which sets out proposals to reduce emissions.</li> <li>The MTS which details policies to reduce transport's contribution to climate change and improve its resilience.</li> </ul> </li> <li>The Mayor has also implemented a series of measures in response to climate change such as the CCZ discount for low emission vehicles.</li> </ul>	<ul> <li>Opportunity for positive publicity campaign to promote public awareness of the benefits for climate change.</li> </ul>
Biodiversity including flora and fauna	•	<ul> <li>Increased nitrogen deposition in the form of NO<sub>x</sub> and NO<sub>2</sub> and elevated NO<sub>x</sub> emissions pose a risk to biodiversity through: <ul> <li>increased nitrogen deposition and overloading by nitrogen favourable species, reducing plant diversity in natural and semi natural ecosystems.</li> <li>increased acidity in soils, limiting species that can survive in such conditions.</li> </ul> </li> <li>ULEZ would result in improvement in NO<sub>2</sub> concentrations of 4.6 ug/m<sup>3</sup> in 2020 and of 2.3 ug/m<sup>3</sup> in 2025 in the CCZ and less towards the Outer Zone.</li> <li>ULEZ would also result in a reduction of NO<sub>x</sub> emissions by just over 50% in the CCZ by 2020 lessening over time and towards the Outer Zone.</li> </ul>	ULEZ would result in a positive effect on nature conservation sites across London (specifically sites in the Hampstead Heath Woods in Camden, Hainault Forest in Redbridge and in Hillingdon) as a result of anticipated borough level decreases in NO <sub>x</sub> emissions.	Positive short- medium term	Moderate	<ul> <li>TfL's TERM identifies opportunities for positive impacts on biodiversity as a result of a reduction in air pollution.</li> <li>The Mayor also has broader policies about biodiversity to make London greener, working to promote living roofs and walls across London.</li> </ul>	<ul> <li>Opportunity for positive publicity campaign to promote public awareness of the biodiversity benefits for the ULEZ</li> <li>Opportunity to promote implementation of complementary policies relating to improving London's air quality.</li> </ul>
Cultural heritage	•	Levels of NO <sub>x</sub> emissions in London pose a threat to cultural heritage assets as a result of pollutants that are principally responsible for causing acid rain. Almost all materials are affected by the deposition of acid, but the degree of damage tends to vary. ULEZ would result in a reduction of NO <sub>x</sub> emissions by just over 50% in the CCZ by 2020 lessening over time and towards the Outer Zone.	ULEZ would reduce the risk of acid rain on cultural heritage assets as a result of NO <sub>x</sub> reductions (particularly within the CCZ) The greatest reductions of NO <sub>x</sub> would be experienced in central London, specifically in the boroughs of City of Westminster and Camden, which together contain a combined total of 3,730 listed buildings, 61 scheduled monuments and 1 World Heritage site. Of these listed buildings, over 350 are categorised as having the highest level of heritage value represented by the Grade 1 status.	Positive long term	Major	• TfL's TERM identifies opportunities for positive impacts on the built environment from a reduction in particulates that can cause discolouration of buildings and NO <sub>x</sub> deposits that leave some building materials more vulnerable to weathering.	<ul> <li>Opportunities to promote implementation of complementary policies relating to improving London's air quality.</li> </ul>

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Cultural heritage	<ul> <li>Soiling is also a threat to cultural heritage assets which is a visual darkening of exposed surfaces by deposition of atmospheric particles.</li> <li>The major contributor to all processes of soiling on buildings is the deposition of PM.</li> <li>The ULEZ would result in reductions in PM<sub>10</sub> emissions between 2020 and 2025 of 16% or 17 tonnes per annum and reductions of PM<sub>2.5</sub> emissions by 19% or 7 tonnes per annum between 2020 and 2025.</li> </ul>	ULEZ would result in a reduced risk of degradation of cultural heritage assets as a result of reduced PM <sub>10</sub> emissions. Most significant in the CCZ (9% or 10tpa in 2020 and 3% or 3tpa in 2025).	Positive long term	Minor	Refer to text on previous page.	Refer to text on previous page.
Water		Not within sco	pe of assessn	nent		
Material resources and waste	<ul> <li>ULEZ would result in a 'phasing-out' of the existing vehicle fleets such as buses and taxis which will result in hazardous vehicle wastes such as: <ul> <li>Vehicle and other oils.</li> <li>Interceptor Wastes.</li> <li>Fuels, brake and anti-freeze fluids.</li> <li>Components, including oil filters.</li> <li>Air bags.</li> <li>Brake pad.</li> <li>Batteries.</li> </ul> </li> <li>Some of these components may be recyclable or recoverable using available technologies at existing facilities. For example, materials such as rare earth elements and precious metals may have economic value that can be realised through resale following recovery.</li> <li>Components that cannot be reused, recycled or recovered will need to be stored at landfill.</li> <li>It is estimated that London will need to identify capacity to manage around 82,000 tonnes of hazardous waste it currently sends to landfill each year plus any increases in arising requiring management in order to achieve the London Plan target of net self-sufficiency.</li> </ul>	ULEZ may result in risk of environmental harm resulting from disposal of hazardous vehicle components from the replaced fleets e.g. batteries. While this risk is considered to be neutral (i.e. the likely scale of impact unable to be determined), it is included in this impact table because of the challenges and opportunities that waste presents. 'While London is recycling more than ever before, more can and should be done for the capital to catch up with the other UK regions and European cities' (GLA, 2014). ULEZ would result in increased pressure on waste management facilities and landfill from replacement of existing vehicle fleets (approximately 22,000 taxis and 50,000 licensed private vehicles operating in London in 2011 and 8,500 buses operating in 2014).	Short- medium term	Neutral Minor	<ul> <li>The Mayor's Business Waste Management Strategy sets out plans to help London's businesses manage their waste more efficiently and effectively. Key parts include:         <ul> <li>Facilitating business support programmes that help businesses understand the financial and commercial opportunities in waste reduction, reuse, recycling and reclaimed materials.</li> <li>Helping businesses to increase their access to recycling services and reducing barriers to waste for reuse, recycling, composting and energy recovery.</li> <li>Providing strategic investment to stimulate the development of new waste management infrastructure within London.</li> <li>Using the planning process in London to drive resource efficiency improvements in the construction and demolition sector.</li> </ul> </li> <li>(Source: GLA, 2014)</li> </ul>	<ul> <li>TfL should consult with the GLA, the Department for Environment, Food and Rural Affairs, the Environment Agency and waste management facility operators on waste management strategies to ensure any TfL policies encourage safe storage, reuse and disposal of hazardous vehicle components as already specified in EU regulations.</li> <li>Investigate and adopt recycling options for vehicle components from replaced existing fleet and encourage private vehicles owners to do the same.</li> <li>For TfL vehicles, utilise existing facilities for disassembling batteries.</li> <li>Opportunity to improve and provid innovative ways of how the industideals with waste.</li> <li>Undertake ongoing monitoring of waste management facilities to assess potential for increased demand on waste/ recycling facilities and landfills and ways of recovering value and reducing amounts of waste sent to landfill.</li> <li>Opportunities to recover rare earth elements from the batteries of single deck buses and zero emission taxis.</li> </ul>
		ULEZ may result in increased demand on electricity and fuel to produce new vehicles (embodied carbon).	Short- medium term	Minor		<ul> <li>Lobby government to assist with encouraging operators of TfL bus to sell and redeploy replaced vehicle fleet to other locations where there are lower concentrations of air pollutants rather than disposal.</li> <li>Opportunity to promote enhancement of existing Climate Chance Mitigation and Energy</li> </ul>
						Strategy which includes lobby for and encouraging sustainal energy sources

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Landscape, townscape and urban realm	<ul> <li>TfL is proposing to utilise their existing comprehensive camera enforcement network along the CCZ boundary to remove the requirement for new cameras for ULEZ.</li> <li>TfL is looking into the ability to alter existing CCZ signs to include information about the ULEZ rather than implement new signs.</li> <li>Londoners' satisfaction with the condition of the local urban realm remains relatively good with the mean rating for satisfaction standing at 74 out of 100.</li> </ul>	ULEZ may result in increased visual clutter (street level) as a result of additional street furniture required for new commercial vehicle rapid charging points.	Short- medium term	Minor	<ul> <li>TfL's <i>Streetscape Guidance 2009</i> provides information on high quality streetscape design including design principles and palette of materials.</li> <li>The Roads Task Force was set up by the Mayor in 2012 to tackle the challenges facing London's streets and roads. The Roads Task Force report, published in July 2013, sets out the vision and direction for London's streets and roads.</li> </ul>	<ul> <li>TfL to support further research in relation to the use of wireless charging technologies (e.g. at bus depots and taxi ranks).</li> <li>Identify optimal charge point locations taking account of the need for sensitive placement of infrastructure to minimise impact on streetscape.</li> <li>Opportunity to work closely with technology providers to ensure new infrastructure meets London's needs and is in keeping with existing streetscape where possible.</li> </ul>
Health and wellbeing	<ul> <li>Air pollution</li> <li>Studies of air pollution have shown that high levels of ambient air pollution are associated with strong increases in adverse health effects.</li> <li>Although ULEZ would operate in central London, it is expected that the air quality and health benefits would extend into London more generally as drivers switch to cleaner vehicles.</li> <li>Without ULEZ, in 2020 63% of people in central London would be living in areas above the NO<sub>2</sub> annual limit value, 12.7% in the Inner Zone and 1.7% in the Outer Zone. While this would reduce in 2025 17.1% of people in central London would still be living in areas above the NO<sub>2</sub> annual limit value, 2.2% in the Inner Zone and 0.2% in the Outer Zone.</li> <li>Key receptors include road users, pedestrian, residential properties, schools, hospitals, the elderly / children, care homes, open spaces, public rights of way and nature conservation sites.</li> <li>Without ULEZ, in 2020 the number of care homes in central London exposed to concentrations exceeding the annual mean NO<sub>2</sub> is 1.</li> <li>Without ULEZ, in 2020 the number of hospitals in central London exposed to concentrations exceeding the annual mean NO<sub>2</sub> is 27.</li> <li>Without ULEZ, in 2020 the number of hospitals in central London exposed to concentrations exceeding the annual mean NO<sub>2</sub> is 29.</li> <li>The health effects associated with short-term (acute) exposure to ambient air pollution include premature mortality (deaths brought forward), respiratory and cardio-vascular hospital admissions, exacerbation of asthma and other respiratory symptoms.</li> <li>Without ULEZ in 2020, 1,448 respiratory admissions would be attributable to particulate exposure.</li> </ul>	<ul> <li>ULEZ would result in increased personal health and wellbeing as a result of improvements to air quality as people switch to less polluting vehicles and other modes of transport e.g. public transport, walking and cycling.</li> <li>ULEZ would result in reductions in the number of people living in areas above the NO<sub>2</sub> annual limit value in 2020 and 2025. Specifically in 2020 ULEZ would result in the following reductions of people living in areas above the NO<sub>2</sub> annual limit value: <ul> <li>Central Zone – reduction of 74%.</li> <li>Inner Zone – reduction of 50%.</li> <li>Outer Zone – reduction of 42%.</li> </ul> </li> <li>A large reduction in the number of care homes, hospitals and schools in areas exceeding the NO<sub>2</sub> AQO is projected across London compared to without ULEZ. This fall is greatest in central London and is as follows: <ul> <li>Care homes – decrease from 1 (without ULEZ) to 0 (with ULEZ).</li> </ul> </li> <li>Hospitals – decrease from 29 (without ULEZ) to 10 (with ULEZ).</li> <li>Schools – decrease from 27 (without ULEZ) to 4 (with ULEZ).</li> <li>In 2020 ULEZ would, as a result of positive health benefits (from the reduction in NO<sub>x</sub>, PM<sub>10</sub> and PM<sub>2.5</sub> under the ULEZ for the GLA area), result in reductions of 4,123 life-years lost across Greater London. However this reduces in 2025.</li> </ul>	Positive long term Positive long term Positive long term	Major Major Major Moderate	<ul> <li>Health benefits from reductions in air pollution and noise as a result of the proposed ULEZ can be enhanced by encouraging the use of public transport, cycling and walking.</li> <li>The MTS promotes walking and cycling and is supported by a number of investments in public transport and walking and cycling including:         <ul> <li>Developing Crossrail to provide a high speed rail service between the east and west of the city.</li> <li>Improving the Tube with extended services, new trains and stations which are easier to navigate.</li> <li>Making London's transport more accessible for people with disabilities and connecting up the city's more deprived areas.</li> <li>1,700 hybrid double deck buses by 2016, including 600 New Routemasters.</li> <li>Revolutionising railways with more routes and faster journeys.</li> <li>Making the most of the River Thames with more pier capacity, more services and a ticketing system which is integrated with the rest of London's transport.</li> <li>Promoting a Cycling Revolution to encourage more journeys to be made by bicycle in London.</li> <li>Making Walking Count, making London an easier city to travel around by foot (GLA, 2014).</li> </ul> </li> <li>TfL's Business Plan 2013 confirms investment in public transport, walking and cycling infrastructure and the Mayor's 2050 Infrastructure Investment Plan details significant investment in public transport and urban realm.</li> </ul>	<ul> <li>TfL should investigate the potential expansion of ULEZ / raising of ULEZ standards in the future (i.e. post 2025) or the introduction of complementary adjacent zones for low emissions neighbourhoods. This could result in further air quality benefits. Dependent on vehicle market, infrastructure, opportunity and compliance costs at the time of investigation and public consultation.</li> </ul>

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Health and wellbeing	<ul> <li>Noise</li> <li>ULEZ would result in increased usage of low and zero emission vehicles, particularly the introduction of zero emission single-decker electric and hydrogen buses and low emission hybrid double-decker buses.</li> <li>The EA concluded that use of hybrid buses can offer a reduction in noise levels of around 3dB(A) or 30% compared to conventional, diesel based engines. Fully electric buses are likely to offer even greater reductions in noise and vibration.</li> <li>Initially, the greatest proportion of routes running low and zero emission buses will be in central London.</li> <li>Additional noise reductions are likely to be achieved through phasing out older taxis, with the ULEZ setting the age limit at 10 years. All newly licensed taxis and private hire vehicles are proposed to be zero emissions capable by 2018.</li> </ul>	ULEZ would result in small health benefits as a result of reduction in noise and vibration annoyance and disruption for some receptors and communities (where overall road traffic noise along some roads decreases as a result of increased usage of low and zero emission vehicles).	Positive short- medium term	Minor	Refer to text on previous page.	• Reductions in noise levels could be enhanced through encouraging increased usage of hybrid, electric and hydrogen buses. Initially the greatest proportion of routes running these buses will be in central London. This could be extended into the inner and outer zones.
	<ul> <li>Active travel</li> <li>The ULEZ would levy a toll on private vehicular traffic that does not meet the required low emission vehicle status.</li> <li>TfL expect ULEZ to reduce car use, promote sustainable travel / mode shift, increase the proportion of ultra-low or zero emission vehicles and stimulate the uptake / development of low emission vehicles.</li> <li>Data from a London Councils poll was used to understand residents' attitudes towards walking in London. It revealed they wanted: better road safety, better condition of pavements, a safe urban environment, a less polluted environment, a quieter urban environment, better information and way finding, more information on health benefits.</li> </ul>	ULEZ may contribute towards the promotion of active travel by providing a less polluted urban environment and therefore creating a better environment for active travel. For those entering the ULEZ who do not have compliant vehicles, the ULEZ may also deter them from entering the zone or result in a modal shift to greener transport modes (e.g. bus, tube, train, cycling).	Positive long term	Likely scale of impact could not be quantified with data available - expected to be minor		None proposed.
	<ul> <li>Crime reduction and community safety</li> <li>Newer vehicles tend to be safer, and are part of the reason why road safety in the UK has improved.</li> <li>CCTV does not have a large effect on reducing the fear of crime and therefore any increased surveillance of the ULEZ will unlikely deter the occurrence of illegal driving or street crime. Regardless, TfL are proposing to use existing CCZ infrastructure where possible.</li> <li>The crime rate per million passenger journeys on TfL's public transport network has decreased between 2004/2005 and 2012/2013 for all transport types (bus, London Underground / Docklands Light Rail (DLR), tramlink and overground).</li> </ul>	ULEZ would result in small improvements in road safety as a result of an increase in newer vehicles on the road.	Short- medium term	Neutral	<ul> <li>The MTS provides proposals to improve safety and security including for improving public transport safety and road safety and reducing crime, fear of crime and antisocial behaviour.</li> <li>Additionally the Mayor has the Safe Streets for London – the Road Safety Action Plan for London 2010 and TfL continues to do a lot of work to encourage newer and safer lorries with, for example, cyclist safety features like side guards. TfL is currently consulting on a Safer Lorries scheme to require all HGVs in London to have side guards and extended view mirrors.</li> </ul>	None proposed.

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Population and equality	<ul> <li>Traffic, air and noise</li> <li>The baseline for assessing potential impact on equality groups from ULEZ in terms of traffic noise and air quality draws heavily on the EA and HIA.</li> <li>The EA concluded that at a zonal level there would be overall reductions in traffic volume across London in 2020 and 2025 as a result of ULEZ.</li> <li>Any changes in daily traffic flows caused by a greater proportion of travellers changing mode or no longer travelling or re-routing as a result of ULEZ would likely to be extremely minor.</li> </ul>	ULEZ would result in a minor long term beneficial reduction in the average exposure to $NO_2$ for all people in 2020 and 2025 however this would be greater for those in deprived areas as the average level of reduction on $NO_2$ concentrations is higher in the most deprived areas.	Positive long term	Minor	<ul> <li>The Mayor has outlined a strategy of measures to tackle pollution:         <ul> <li>Cleaning up London's bus fleet.</li> <li>Capping the age of London's taxi and private hire fleet.</li> <li>Setting new and tighter standards for the London LEZ.</li> <li>Encouraging the update of electric and other low emission vehicles.</li> <li>Investing record amount in cycling (TfL, 2014).</li> </ul> </li> </ul>	None proposed.
	<ul> <li>In all groupings of Lower Super Output Areas (i.e. at all levels of deprivation) ULEZ would bring about a reduction in NO<sub>2</sub> concentrations.</li> <li>Over 26% of London falls within the most deprived 20% of England.</li> <li>Without ULEZ those in the most deprived parts of London are expected to live in areas with higher annual mean NO<sub>2</sub> concentration in both 2020 and 2025.</li> <li>Schools, hospitals and care homes are considered to be particularly sensitive as they house a high density of potentially vulnerable people.</li> <li>In central London in 2020, without ULEZ, it is forecast that 20% of care homes, over 50% of hospitals and 34% of schools will be in areas which experience exceedances of NO<sub>2</sub> AQOs.</li> </ul>	<ul> <li>ULEZ would have a positive differential impact on school age children, older people and pregnant women. This is as a result of the reduction of sensitive receptors (schools, care homes and hospitals) that would be in areas which experience exceedances in NO<sub>2</sub> emissions. Specifically, in central London, ULEZ would result in</li> <li>0% of care homes being in areas of exceedance (compared with 20% without ULEZ)</li> <li>17.5% of hospitals being in areas of exceedance (compared with 50% without ULEZ)</li> <li>5% of schools being in areas of exceedance (compared with 34% without ULEZ)</li> <li>The positive impacts would be major in central London, moderate in inner London and minor in outer London.</li> </ul>	Positive long term	Moderate	<ul> <li>The Mayor is also supportive of:</li> <li>The Government to help replace the most polluting diesel cars with lower emission vehicles and giving drivers cash incentives to switch to cleaner vehicles.</li> <li>A new public awareness programme called Breathe Better Together to tackle air pollution at schools and an ambitious public awareness campaign to be launched in October (GLA, 2014).</li> </ul>	
	<ul> <li>Cars / motorcycles</li> <li>The ULEZ will require some operators of cars and motorcycles to upgrade their vehicles to comply with the ULEZ emission standards or otherwise will be required to pay the ULEZ charge.</li> <li>Those residents within the proposed ULEZ will benefit from TfL's committed mitigation.</li> <li><i>TfL has committed to providing mitigation of the impact on all residents living in the ULEZ area, who will be granted a three year time limited 100% discount to recognise that they are unable to avoid the ULEZ area and may require more time to change their vehicle for one to meet ULEZ emissions standards.</i></li> <li>People on lower incomes are potentially more likely to work unsocial hours with more reliance on travel by car.</li> <li>As noted in the EBIA, more cars enter the proposed ULEZ between 19:00 and 22:00 than between 7:00 and 10:00. While there is no breakdown of journey purpose by time of day it can be assumed that some of these people will be working in the night time economy and benefiting from no Congestion Charge.</li> <li>Unlike the Congestion Charge which is only payable between 7:00 and 18:00 the ULEZ charge is payable 24 hours.</li> </ul>	ULEZ may have differential impact on low income workers who work more unsocial hours and travel to work in central London by car. This is because they may be unable to afford a ULEZ compliant vehicle or pay the charge. This impact would be offset by complementary policies which work towards improved night time services for London's public transport system.	Short- medium term	Minor	<ul> <li>The MTS promotes alternative modes of travel (to the car) by a number of investments in public transport and walking and cycling.</li> <li>Some of these relate to improving the Tube with extended services (including 24 hours services on some lines) and new trains and stations.</li> <li>TfL also provide good coverage and frequency of night bus services.</li> </ul>	<ul> <li>TfL should undertake targeted communication and awareness raising with affected groups.</li> <li>TfL should continue existing work on improvements to the coverage and frequency of night bus services and later London underground services.</li> </ul>

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Population and equality	<ul> <li>Cars / motorcycles (continued)</li> <li>For the CCZ, disabled people in the UK meeting the qualification criteria are eligible for the Blue Badge scheme which helps the holder park (onstreet) close to a destination as a driver or passenger. Some organisations also qualify e.g. charities on a discretionary basis if they transport people with such disabilities.</li> <li>3.1% of the London population are Blue Badge holders. These vehicles represent 5-7% of all movements into / out of the CCZ.</li> <li>In 2011, the average age of a blue badge registered petrol vehicle entering the CCZ was eight years. Assuming the age profile of vehicles is the same in 2020, approximately 16% of all petrol blue badge registered vehicles could be non-compliant when ULEZ comes into operation.</li> <li>In 2011, the average age of a blue badge diesel vehicle was 5 years. Assuming the age profile of vehicles is the same in 2020, approximately 16% of all petrol blue badge registered vehicles who enter the zone do so less than once per month and therefore are less likely to be affected by the proposed ULEZ. The average number of trips into the CCZ (curing charging hours) is 2 per month for vehicles that do enter the zone.</li> <li>A proportion of Blue Badge holders will require vehicles supplied through the Mobility scheme through which they receive VAT relief on substantially and permanently adapted vehicles.</li> <li>Disable residents living in the ULEZ area, who will be granted a three year time limited 100% discount to recognise that they are unable to avoid the ULEZ erea and may require more time to change their vehicle for one to meet ULEZ emissions standards.</li> </ul>	It is not anticipated that there will be a disproportionate effect on disabled car drivers. However, it may be more difficult for disabled persons to find alternative modes of accessible transport to central London. This impact may be offset by TfL's research and work to improve accessibility for all to London's transport.	Short- medium term	Minor	<ul> <li>Recent improvements across the London transport network (as identified in TfL's Understanding the travel needs of London's diverse communities) include:         <ul> <li>'Turn up and go' on all London Overground stations, so that disabled people needing assistance can arrive at stations and have staff help them without needing to book.</li> <li>Roll out of accessible boarding ramps at many Tube stations.</li> <li>Staff training, particularly for bus drivers to address concerns raised by stakeholders such as wheelchair users.</li> <li>Programmes to improve safety and security when travelling in London, such as Project Guardian which aims to reduce incidents of unwanted sexual behaviour across the transport network (TfL, 2014).</li> </ul> </li> <li>TfL's Single Equalities Scheme sets out goals and activity to remove barriers to travel in London wherever possible. It focuses on delivering:         <ul> <li>A transport system that is safe and reliable for all.</li> <li>Improved physical accessibility across the network.</li> <li>Affordable transport.</li> <li>Engagement with passengers and stakeholders from London's communities.</li> <li>Accessible information.</li> <li>A workspace that is representative of London's diverse communities (TfL, 2014).</li> </ul></li></ul>	None proposed.

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
Population and equality	<ul> <li>Non-TfL buses and coaches</li> <li>The EqIA focused on those services or journey types which are specifically provided for people with protected characteristics or socio-economically deprived communities including: <ul> <li>School buses for pupils who reside outside the ULEZ but attend a school located in the ULEZ (on a daily basis).</li> <li>School trips for pupils in schools outside of the ULEZ who travel into the ULEZ for educational purposes (likely to be very infrequent in most cases).</li> </ul> </li> <li>There would be a financial cost to vehicle operators in adapting to use ULEZ compliant vehicles from 2020 or from the need to pay the ULEZ charge if the vehicles are non-compliant.</li> <li>It is assumed larger operators would have the ability to move vehicles operate in the ULEZ.</li> <li>Coaches will be used for educational and leisure trips into London by schools from across the UK and additional costs associated with complying with the ULEZ could at a worse case be passed onto local authorities and/or to individual children travelling.</li> <li>Most schools would hire coaches rather than own them and it is anticipated that schools will have the option of hiring from coach operators that operate ULEZ emissions compliant vehicles to avoid incurring charges from using non-compliant vehicles.</li> </ul>	Any increase of the costs of school trips by private hire bus or coach to central London would have a differential effect on those children from low income families. This impact may be offset by complementary policies which work towards improvements to London's public transport system. Schools could use alternative modes of transport for school trips.	Short- medium term	Minor	<ul> <li>School children receive free transport on TfL services (bus, underground, overground, DLR, Tramlink).</li> <li>The MTS identifies a number of investments in public transport (which may be used as an alternative mode of transport for school trips including):         <ul> <li>Developing Crossrail to provide a high speed rail service between the east and west of the city.</li> <li>Improving the Tube with extended services, new trains and stations which are easier to navigate.</li> <li>Making London's transport more accessible for people with disabilities and connecting up the city's more deprived areas.</li> <li>1,700 hybrid double deck buses by 2016, including 600 New Routemasters.</li> <li>Revolutionising railways with more routes and faster journeys.</li> <li>Making the most of the River Thames with more pier capacity, more services and a ticketing system which is integrated with the rest of London's transport (GLA, 2014).</li> </ul> </li> <li>TfL's Business Plan 2013 confirms investment in public transport infrastructure and the Mayor's 2050 Infrastructure lnvestment Plan details significant investment in public transport and urban realm.</li> </ul>	<ul> <li>Engage in early and proactive communication of the ULEZ.</li> <li>TfL to work with manufacturers and government to identify ways of retrofitting coaches and provide financial support to operators for adoption of the technology.</li> </ul>

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in a
Population and equality	<ul> <li>Taxis and PHVs</li> <li>ULEZ introduces a requirement for all newly licensed taxis from 2018 to be zero emission capable.</li> <li>The rolling age limit for conventional non zero emission capable diesel taxis will be reduced from 15 years to 10 years. TfL has committed to the following mitigation should this reduction in the rolling age limit be taken forward.</li> </ul>	Lowering the taxi age limit may have a disproportionate effect on the third of licensed taxi drivers who are older (60+) who may choose to retire early rather than upgrade to a ULEZ compliant vehicle. This impact would be offset by TfL's committed mitigation to establish a specific fund for drivers of older taxis to help them switch to newer vehicles as well as the plug-in car and van grants from the Office for Low Emission Vehicles (OLEV), however the take up of the fund is unknown.	Short- medium term	Minor	• The plug-in car and van gran provides a financial incentive UK address that need to repl become compliant with ULE2 this is a grant of 25% towards vehicle up to a maximum of £ grant is only guaranteed until have reserved the option to r value in 2017 or once 50,000
	TfL will establish a specific fund for drivers of older taxis to help them switch to newer vehicles. It is anticipated that grants would be offered to eligible taxi owners and that the proposed scheme would be phased from 2018 to smooth the impact of a reduced age limit in 2020.				whichever comes sooner.
	• The age profile of the taxi fleet indicates that about one-third will require replacement (many with zero emission vehicles from 2018).				
	• ULEZ will require operators of PHVs to upgrade their vehicles to comply with the ULEZ emission standards or otherwise they will be required to pay the ULEZ charge.				
	• From 2018 new PHV licences will no longer be granted to non-ULEZ compliant vehicles (i.e. Euro 6 diesel, Euro 4 petrol); and any newly licensed PHV under 18 months old will need to be zero emissions capable.	BAME are disproportionately represented as PHV drivers and therefore any additional costs from ULEZ may impact upon this group disproportionately.	Short- medium term	Minor	
	Generally taxis operate predominantly in inner London, whereas PHVs trips are evenly distributed across Greater London.	This impact would be offset by the plug-in car and van grants from the OLEV, however the take up of the fund is unknown.			
	• 40% of PHV fleet never enters central London.				
	65% of PHV fleet turnover are second hand vehicles.				
	• Taxis are principally a central London transport mode with 84% of all taxi trips taking place within, to or from central London and 30% beginning and ending within it.				
	• The driver profiles of taxis and PHVs are very different both in terms of age and ethnicity.				
	• Taxi drivers are older and predominantly white males; PHV drivers tend to be younger and the majority are of BAME origin.				
	• The impact of ULEZ will depend on the extent to which it will change the overall supply and geographical patterns of operation by both modes.				
	• Although a large part of the PHV market is comprised of minicabs services there are a number of other sectors e.g. executive / chauffer services community transport, school runs etc.				
	that are licensed as PHV services in London.				

### additional to ULEZ)

rants from the OLEV
ive for businesses with a eplace vehicles to
.EZ standards. For a car, and the cost of the of £5,000. However, this ntil 2020 and OLEV
o review the car grant
000 cars have been sold,

#### Opportunities for enhancement or mitigation suggested for further investigation by TfL

- TfL should continue to lobby for the extension of the existing OLEV grant beyond 2017.
- TfL has submitted a bid to OLEV towards grants for the purchase of new ZEC taxis and PHVs.
- TfL is also seeking support from OLEV for investment in supporting infrastructure (e.g. taxi charging points).
- TfL to continue working with the taxi industry to establish likely take up rates of a proposed financial incentive scheme for taxi replacement.

- Prepare easy to understand information on the likely cost of compliance for different types of PHV vehicles; this would be helpful as a means of engaging with PHV companies and licenced drivers to enable them to plan for the introduction of ULEZ.
- Engage in early and proactive communication with industry and ongoing monitoring of compliance in run up to 2020.

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
IIA topic Population and equality	<ul> <li>Baseline for assessment</li> <li>Taxis and PHVs (continued from previous page) <ul> <li>Although a large part of the PHV market is comprised of minicabs services there are a number of other sectors e.g. executive / chauffer services community transport, school runs etc. that are licensed as PHV services in London.</li> <li>The TfL report Understanding the travel needs of London's diverse communities (August 2014) found that in comparison with men, London's women had concerns over crime/antisocial behaviour on public transport. Further personal safety 'fear of intimation or abuse' is a barrier to travel for Lesbian, Gay, Bisexual and Transgendered (LGBT) community.</li> </ul> </li> </ul>	Positive and negative impacts of the ULEZ         ULEZ may result in a reduction in supply of fully accessible taxis / adapted PHVs.         This impact would be offset by complementary policies which work towards improvements to London's public transport system.         ULEZ may have a differential effect on women and the LGBT community arising from increased fear for personal safety in central London and other town centres in Greater London at night as a result of a potential decrease of available taxis.         This impact would be offset by complementary policies which work towards safety improvements to London's public transport system.			<ul> <li>Complementary policies (in additional to ULEZ)</li> <li>There is a subsidised taxi service, funded by TfL and London boroughs (Taxicard) for people who have mobility impairments or who cannot easily use other public transport modes. The number of London Taxicards currently in circulation is 75,545 members and there were 350,000 Taxi card trips during 2012/13.</li> <li>The Mayor has a 3 year strategy, <i>The Right Direction</i>, to improve transport safety and security including increasing confidence in the safety and security of travelling in London.</li> <li>The MTS promotes measures to improve the physical accessibility of the transport system, including streets, bus stops, stations and vehicles. It also provides proposals to improve safety and security including for: <ul> <li>Improving public transport safety.</li> <li>Improving road safety.</li> <li>Reducing crime, fear of crime and antisocial behaviour.</li> </ul> </li> <li>Recent improvements across the London transport network (as identified in TL's <i>Understanding the travel needs of London's diverse communities</i>) include: <ul> <li>'Turn up and go' on all London Overground stations, so that disabled people needing assistance can arrive at stations and have staff help them without needing to book.</li> <li>Roll out of accessible boarding ramps at many Tube stations.</li> <li>Staff training, particularly for bus drivers to address concerns raised by stakeholders such as wheelchair users.</li> <li>Programmes to improve safety and security when travelling in London, such as Project Guardian which aims to reduce incidents of unwarted sexual behaviour across the transport network (TfL, 2014).</li> </ul> </li> <li>TifL's Single Equalities Scheme sets out goals and activity to remove barriers to travel in London wherever possible. It focuses on delivering: <ul> <li>A transport system that is safe and reliable for all.</li> <li>Improved physical accessibility across the network.</li> </ul> </li> </ul>	mitigation suggested for further

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies (in additional to ULEZ)
Population and equality	<ul> <li>Vans and minibuses</li> <li>Around 900 minibuses enter the CCZ 50 times a year or more and account for nearly 60% of all entries into the zone for this vehicle type.</li> <li>The ULEZ will require operators of vans to upgrade their vehicles to comply with the ULEZ emission standards or otherwise will be required to pay the ULEZ charge.</li> <li>Euro 6 diesel vans and minibuses will not be available until 1 September 2016, four years prior to operation of ULEZ.</li> <li>The ULEZ charge is £12.50 which is lower than the equivalent charge for LEZ which is £100 in order to mitigate costs of compliance.</li> <li>Van ownership is broadly split 50:50 between companies and private owners showing the importance of vans to owner run businesses.</li> </ul>	The costs of compliance associated with ULEZ may have a disproportionate impact on BAME businesses using vans in central London due to their disproportionate representation in the retail and wholesale industry in Greater London; a sector which makes high use of this type of vehicle. This impact would be offset by the plug-in car and van grants from the OLEV, however the take up of the fund is unknown.	Short- medium term	Minor	<ul> <li>The plug-in car and van grants from the OLEV provides a financial incentive for businesses with a UK address that need to replace vehicles to become compliant with ULEZ proposals. For a van this is a grant of 20% towards the cost of the vehicle up to a maximum of £8,000. However, this grant is only guaranteed until 2020 and OLEV have reserved the option to review the car grant value in 2017 or once 50,000 cars have been sold, whichever comes sooner.</li> <li>TfL commissioned some research to investigate the likely impact of the ULEZ on independent businesses, particularly BAME owned/managed businesses. This research found that the introduction of ULEZ would not have a greater impact on establishments that are BAME owned/managed.</li> </ul>
	<ul> <li>Evidence indicates that London's BAME community are disproportionately represented as business owners in the wholesale and retail business; a sector which has high levels of van use.</li> <li>The total BAME population in the ULEZ area is predicted to increase from 30% in 2011 to 33% in 2025, together with a corresponding decrease in white ethnicity.</li> <li>'Dial a Ride' is a service directly operated by TfL providing door to door transport for older and disabled people in London who cannot always use buses, trains or the Tube. It mainly uses minibuses of eight seats, with a few larger vehicles. Costs associated with compliance will be borne by TfL.</li> </ul>	Increased cost of access to central London by minibus may have differential impact on those groups reliant on charitable or voluntary services (e.g. disabled, older people, faith groups).	Short- medium term	Minor	• There is a subsidised taxi service, funded by TfL and London boroughs (Taxicard) for people who have mobility impairments or who cannot easily use other public transport modes. The number of London Taxicards currently in circulation is 75,545 members and there were 350,000 Taxi card trips during 2012/13.

to th t of 📍 bа is ve nt

## Opportunities for enhancement or mitigation suggested for further investigation by TfL

- TfL should continue to lobby for the extension of the existing OLEV grant beyond 2017.
- Opportunity for TfL to work with SME business representatives in retail and wholesale sectors in order to identify potential measures which could help to mitigate the anticipated impact.
- Investigate the feasibility of establishing consolidation centres on the edge of the proposed ULEZ with goods being transferred to low emission vehicles for onward movement into the ULEZ.
- Engage in early and proactive communication of the ULEZ.
- TfL should secure money in their \_ • business plan to assist dial-a-ride fleet where necessary.

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policies ULEZ)
London's economic competitive- ness	<ul> <li>Air quality</li> <li>There is a health 'burden' associated with absolute levels of pollutant concentrations including for chronic mortality, respiratory and cardiovascular hospital admissions.</li> </ul>	ULEZ may provide for London to become a more attractive city for business and tourists as a result of improvements to air quality and subsequent health impacts.	Positive long term	Minor	<ul> <li>The Mayor's Economi Strategy sets out his v to the London econom be realised.</li> </ul>
	<ul> <li>The central value of these 'burdens' are £35,000 for chronic mortality, £2,600-£10,700 for respiratory hospital admission and £3,000-£9,900 for cardiovascular admissions.</li> <li>HGVs</li> <li>Retail and wholesale distribution, and construction are the main sectors served by HGVs.</li> <li>It is anticipated that the vast majority of businesses that have HGVs regularly entering the proposed ULEZ will continue to do so with minimal impact on the sector or on London's economy as a whole.</li> <li>LGVs</li> </ul>	The health benefits from ULEZ will result in an economic benefit associated with reductions in air pollution. The valuation of health improvement captures a number of economic impacts, including direct impact on the utility of the affected individual, reduction in medical costs and increase in productivity. The improved health outcomes arising from the reduction in NO <sub>X</sub> , PM <sub>10</sub> and PM <sub>2.5</sub> under the ULEZ for the GLA area are estimated to have a total monetised benefit of £101m in 2020 and £32m in 2025.	Positive long term	Moderate	
	<ul> <li>LGVs are used across all sectors of London's economy, from servicing financial and business service companies to supporting independent retailers and food outlets.</li> <li>A relatively small proportion of LGVs will be compliant without further investment by operators.</li> <li>Cars</li> <li>People who travel to work by car usually do so outside of normal commuting hours. It is assumed that some of these people benefit from free parking and no Congestion Charge.</li> <li>Non-TfL buses, coaches and minibuses</li> <li>15,000 vehicles entering the proposed ULEZ in 2014 were minibus type vehicles operating: scheduled services; inter-company shuttles for multi-site operators and airport-hotel link services; and services relating to private hire and private uses (by schools, and clubs).</li> <li>25,000 vehicles entering the proposed ULEZ in 2014 were buses and larger coaches.</li> <li>Taxis</li> <li>There are more than 22,000 licensed taxis in London and nearly 25,000 taxi drivers, the vast majority of which operate within central London.</li> <li>84% of all taxi trips take place within, to or from central London; 30% beginning and ending within it. On an average day, about 185,000 passenger-carrying taxi journeys are made carrying 278,000 passengers.</li> <li>It is estimated that around a fifth of taxi passenger trips are made by overseas visitors and a further fifth by domestic visitors to London.</li> <li>PHVs</li> <li>PHVs are split into minicab operators and chauffeur / executive services.</li> <li>Minicabs are estimated to carry around 230,000 passengers a day and chauffeur/executive services a further 50,000 aday.</li> <li>Minicabs are mostly used in outer London and executive services are more dispersed, although there is a high proportion of airport related trips.</li> </ul>	In the first year of operation, ULEZ may result in a loss of 0.03-0.08% to the London economy broken down as follows: • 0.4% to the retail sector • 0.4% to the construction sector • 0.4% to the accommodation / catering sector • 1-2% to the night time economy • 1% to the coach sector • 0.2% to the tourist sector This impact will decline quickly as levels of compliance increase. Further some of the cost of compliance increase. Further some of the cost of compliance (e.g. vehicle replacement and retrofitting) will be spent with other London businesses so it is not a total loss to the London economy while some operators impacted are not based in London so the net impact on London's economy will be less than this figure. In addition operators that purchase new vehicles should experience reduced operating and maintenance costs. In future years the cost will fall as a higher proportion of vehicles become compliant, so that by 2025 the cost will reduce to virtually zero with the exception of LGV operators.	Short- medium term	Minor	<ul> <li>The Mayor's Transpor supported by a numbe public transport and w</li> <li>The plug-in car and va OLEV provides a finar businesses with a UK to replace vehicles to with ULEZ proposals. grant of 25% towards vehicle up to a maxim for a van this is a gran the cost of the vehicle of £8,000. However, tf guaranteed until 2020 reserved the option to grant value in 2017 or have been sold, which sooner.</li> </ul>

eies (in additional to Z)	Opportunities for enhancement or mitigation suggested for further investigation by TfL
omic Development is vision with respect nomy and how it can	<ul> <li>Opportunity for positive publicity campaign to promote public awareness of the benefits for the ULEZ.</li> </ul>
	None proposed.
port Strategy is nber of investments in d walking and cycling. d van grants from the nancial incentive for UK address that need to become compliant als. For a car, this is a ds the cost of the kimum of £5,000 and rant of 20% towards cle up to a maximum r, this grant is only 120 and OLEV have n to review the car or once 50,000 cars nichever comes	<ul> <li>TfL to work with manufacturers and government to identify ways of retrofitting coaches and provide financial support to operators for adoption of the technology.</li> <li>Investigate further ways of working with the Government to provide financial assistance to operators of vehicles to replace non-compliant vehicles with compliant vehicles, including support for the Mayor's call for a diesel scrappage scheme and investigation of its application to HGVs.</li> </ul>
	<ul> <li>Continue improvements to the coverage and frequency of night bus services and later London underground services throughout the week.</li> <li>Raise awareness of the availability of night time public transport services.</li> <li>TfL should lobby for the extension of the existing OLEV grant beyond 2017/2018.</li> </ul>

IIA topic	Baseline for assessment	Positive and negative impacts of the ULEZ	Duration of impact	Scale of impact	Complementary policie ULEZ)
Small to medium sized enterprises	<ul> <li>HGVs</li> <li>It is envisaged that between 10-20% of non-compliant HGVs that regularly enter London will be replaced by bringing forward purchase decisions by 12 months.</li> <li>Increased costs would be incurred by businesses across London, the South East and, to a lesser extent, the rest of the UK.</li> <li>The ability of operators to pass on costs to customers would depend on the proportion of compliance within the sector and the degree of competition between operators.</li> <li>It is estimated that 95% of fleet operators with more than 10 vehicles registered may be able to reallocate vehicles to ensure that only compliant vehicles enter the ULEZ. For smaller fleet operators, it is estimated that this falls to around 75%.</li> <li>LGVs</li> <li>It will not be possible to retrofit non-compliant LGVs to meet the proposed ULEZ requirements. LGV operators can either buy new vehicles or switch to second hand petrol vehicles.</li> <li>A proportion of local commuter services, sight-seeing vehicles and tourist coaches will not be compliant by 2020.</li> </ul>	The total costs to businesses of either complying with the proposed ULEZ or paying the charge is expected to be around £120-250m in the first year which will fall disproportionately on SMEs but will diminish over time as the proportion of vehicles becoming compliant increases.	Short- medium term	Minor	<ul> <li>The plug-in car and v OLEV provides a finat businesses with a UH to replace vehicles to with ULEZ proposals grant of 25% towards vehicle up to a maxim for a van this is a gra the cost of the vehicle of £8,000. However, guaranteed until 202/ reserved the option to grant value in 2017 of have been sold, whice sooner.</li> <li>The Mayor's TERM p the Mayor's Air Quali fund of £6m from 201 2015/2016 with expe continue to £20m. It boroughs to apply to Boroughs and use fu delivering innovative improvement projects could be to help esta centres and promote</li> </ul>
	<ul> <li>10-30% of non-compliant vehicles that regularly enter the proposed ULEZ may be replaced by bringing forward purchase decisions by up to 24 months.</li> <li>Given the high proportion of non-compliant LGVs, thereby reducing competition between operators, operators may be able to pass additional costs on to customers.</li> </ul>	ULEZ may result in a reduction in competition between, and diversity of, SMEs as a result of the cost of around 0.03-0.08% of the annual value of London's economy of £300bn in the first year of operation.	Short- medium term	Minor	
	Coaches				
	• A proportion of local commuter services, sight-seeing vehicles and tourist coaches will not be compliant by 2020.				
	• 10-30% of non-compliant vehicles that regularly enter the proposed ULEZ may be replaced by bringing forward purchase decisions by up to 24 months.				
	• Some fleet operators will also be able to reallocate vehicles to ensure compliance.				
	Taxis				
	• If the reduction in the maximum age of non zero emission capable taxis from 15 to 10 years, is taken forward with ULEZ, around a third of taxis will need to be replaced sooner than required.				
	PHVs				
	• The majority of minicab trips do not enter the proposed ULEZ and large fleet operators may have some flexibility in moving vehicles around.				
	• Other PHV operators, in particular tour guides and those who operate contracts for local authorities, may use different types of vehicles to those commonly used for minicab purposes given the nature of the work they do.				

#### cies (in additional to Z)

d van grants from the nancial incentive for UK address that need to become compliant als. For a car, this is a rds the cost of the kimum of £5,000 and grant of 20% towards icle up to a maximum r, this grant is only 020 and OLEV have n to review the car 7 or once 50,000 cars nichever comes

A provides details on ality Fund, an initial 013/2014 to bectations for this to It provides for local to become Cleaner Air funding towards ve air quality cts. One example tablish consolidation te alternative vehicles.

#### Opportunities for enhancement or mitigation suggested for further investigation by TfL

- TfL should continue to lobby for the extension of the existing OLEV grant beyond 2017/2018.
- Opportunity to identify and work with representatives (e.g. BIDs) of SMEs in those sectors of the economy which are most likely to be impacted in order to identify potential measures to help mitigate this impact.
- Examine the feasibility of establishing consolidation centres on the edge of the proposed ULEZ with goods being transferred to low emission vehicles for onward movement into the ULEZ.
- TfL could support more initiatives like Plugged in Fleets Initiatives (PiFi) which provided consultancy advice to businesses on switching to low emission electric or PHEVs.
- TfL could raise awareness of options other than vehicle ownership such as other forms of freight transport and van sharing clubs.



## 6 Meeting IIA Objectives

6.1.1 Table 6-A summarises how the impacts of the proposed ULEZ will contribute towards each of the IIA objectives.

IIA Topic	IIA Objective	How is it met?
Air quality	To contribute to a reduction in air pollutant emissions and compliance with EU limit values	ULEZ would result in a reduction of the number of sensitive receptors currently exposed to NO <sub>2</sub> exceedances with almost 18,000 properties no longer being exposed. The majority of these (almost 10,000) are in the ULEZ area.
Noise	To reduce disturbance from general traffic noise	ULEZ would result in a reduction in noise levels. This is mainly as a result of the introduction of zero emission single-decked electric and hydrogen buses and low emission hybrid double- decker buses. Noise from these vehicles will reduce by approximately 3dB(A) or 30% compared to conventional diesel based engine buses. This will be supported through replacement of old taxi fleet with zero emission capable vehicles. On its own a reduction of 3dB(A) for every hybrid bus would not represent a major improvement in noise across London. However, the cumulative impact of all the buses on these routes put together (anticipated to be 1,700 hybrid buses by 2016) could be substantial.
Climate change	To reduce CO <sub>2</sub> emissions and contribute to the mitigation of climate change	With ULEZ the total road traffic CO <sub>2</sub> emissions in London would reduce by 123,000 tonnes per annum in 2020 (from the baseline of 6.4 million tonnes per annum) and 169,000 tonnes per annum in 2025 (from the baseline of 6.37 million tonnes per annum). This equates to overall reductions of 2 per cent in 2020 and 3 per cent in 2025.
		Whilst reductions of 2 per cent of total transport emissions in 2020, and 3 per cent in 2025, are relatively small when compared to London's forecast total transport CO <sub>2</sub> emissions, the primary objective of the ULEZ is to improve air quality. Therefore, any contribution towards the Mayor's target of reducing total emissions in London by 60% against 1990 levels by 2025 should be regarded as positive.
		It is anticipated that total transport sector $CO_2$ emissions will need to be in the range of 5.3 to 4.6m tonnes in 2025 to meet the Mayor's target.



IIA Topic	IIA Objective	How is it met?
Biodiversity including flora and fauna	To protect and enhance the natural environment, including biodiversity, flora and fauna	ULEZ would result in a positive effect on a variety of habitats and nature conservation sites, and subsequently habitats residing in conservation sites across London (specifically sites in the Hampstead Heath Woods in Camden, Hainault Forest in Redbridge and in Hillingdon). This is as a result of decrease in NO <sub>x</sub> emissions (by tonnage and percentage). Decreases in NO <sub>x</sub> emissions can help to reduce dominance by species which favour nitrogen overloading. It can also help lower rates of acidification which limits the number and type of species which can survive in such conditions.
Cultural heritage	To protect and enhance historic, archaeological and socio-cultural environment	There will be a reduced risk of acid rain on cultural heritage assets as a result of $NO_x$ reductions (particularly within the CCZ).
Water	To protect and enhance river spaces and waterways through planning and operation	Not applicable – outside of scope
Material resources and waste	To promote more sustainable resource use and waste management	ULEZ could help to achieve the IIA objective of promoting more sustainable resource use and waste management if the material inputs of the new fleets are sourced from recycled or recovered materials where possible. This could help to support development of waste management and recycling facilities in London which may resolve any future issues surrounding the management of waste products from the replaced vehicle fleets.
Landscape, townscape and urban realm	To protect and enhance the built environment and streetscape	<ul> <li>ULEZ would result in some change to London's landscape and streetscape due to:</li> <li>the implementation of infrastructure required to support changing vehicle fleet composition; and</li> </ul>
		<ul> <li>the supporting transition to low and zero emission vehicles through introduction of new commercial vehicle rapid charging points.</li> </ul>
		These changes are likely to be minimal as wireless charging technology becomes available. Any potential negative streetscape impacts of new signage can be mitigated through sensitive design and integration of signage with CCZ.
Health and wellbeing	To contribute to enhanced health and wellbeing for all within London	ULEZ would promote improved personal health and wellbeing as a result of:
		<ul> <li>improvements to air quality as people switch to less polluting vehicles;</li> </ul>
		<ul> <li>increased use of public transport and walking and cycling as an alternative mode of transport (mode shift from car as a result of the ULEZ); and</li> </ul>
		• decrease in the use of private cars for short journey trips in central London.
		Additionally, a large reduction in the number of care homes, hospitals and schools in exceedance areas for $NO_2$ is projected across London as a result of ULEZ. This fall is greatest in the central Zone and is as follows:



IIA Topic	IIA Objective	How is it met?
		• Care homes – decrease from 1 (without ULEZ) to 0 (with ULEZ).
		Hospitals – decrease from 29 (without ULEZ) to 10 (with ULEZ).
		• Schools – decrease from 27 (without ULEZ) to 4 (with ULEZ).
Population and equality	To enhance equality and social inclusion	ULEZ promotes equality and social inclusion through the significant improvements in air quality it will provide across London's communities, as well as reductions in noise in central and inner London through the introduction of quieter vehicles.
		In terms of its effect on access to the transport network ULEZ will have a largely neutral impact, on most people. However, there are a small number of potential differential or disproportionate impacts on particular equality groups within London. TfL would address these impacts through their extensive surveys, research and consultation that they undertake to understand the way people travel, what the barriers to travel are and what they can do to address any issues raised. TfL has developed a Single Equality Scheme which sets out their goals and activity to remove barriers to travel in London wherever possible for all (TfL, 2014).
London's economic competitiveness	Provide an environment which will help to attract and retain internationally mobile businesses	ULEZ would result in improvements to air quality and subsequent health benefits (refer to HIA and EA for detail). Offsetting the cost on businesses and consumers are the economic benefits arising from improved health and reduced impacts on the NHS. The improved health outcomes arising from the reduction in NO <sub>X</sub> , PM <sub>10</sub> and PM <sub>2.5</sub> under the ULEZ for the GLA area are estimated to have a total monetised benefit of £101m in 2020 and £32m in 2025.
		Improved air quality would also make central London a pleasanter place to work, live and visit. The impact on visitor numbers of this benefit cannot be quantified but it is notable that Beijing (albeit with far greater problems than London) reported last year a significant decline in tourist numbers due to poor air quality (Associated Press 2013) and shows that air quality is a factor for people deciding which locations to visit
SMEs	Support the growth and creation of SMEs	ULEZ may result in a reduction in competition between, and diversity of, SMEs as a result of the cost of around 0.03-0.08% of the annual value of London's economy of £300bn in the first year of operation. This impact will be offset by costs of compliance being spent with other London businesses so it is not a total loss to the London economy. Additionally some operators impacted are not based in London so the net impact on London's economy will be less than this figure.
		TfL can further offset impacts on SMEs and achieve this IIA objective through existing complementary policies and suggested mitigation measures (refer to Table 5-A).

Table 6-A Summary of ULEZ against IIA objectives



7

## Summary and Further Recommendations

7.1.1 As shown in Table 5-A the assessment results also show the ULEZ will result in many positive impacts, particularly in terms of the ULEZ contribution towards environmental and health objectives. There are no major or moderate negative impacts identified. Some minor negative impacts have been identified, however these can be appropriately mitigated as detailed above.

## 7.2 Positive impacts of ULEZ

- 7.2.1 Overall the assessment concludes that the ULEZ will make a strong and lasting positive contribution to London's environment and the health and wellbeing of those who live, work and visit it. Without ULEZ, London (and central London in particular) is forecast to experience an ongoing decline in air quality, primarily as a result of vehicle emissions and the associated implications for public health.
- 7.2.2 The positive impacts resulting from ULEZ are in contrast to the do-nothing scenario and are summarised as follows.

#### (a) Environment

- 7.2.3 ULEZ would lead to improvements in air quality particularly in central London.
- 7.2.4 A substantial number of sensitive receptors would no longer be exposed to exceedances of the NO<sub>2</sub> Air Quality Objective as a result of the ULEZ.
- 7.2.5 ULEZ would result in direct reductions in CO<sub>2</sub> emissions through increased uptake of low and zero emission vehicles and greater compliance with more stringent EURO fuel standards.
- 7.2.6 There may be further indirect reductions of CO<sub>2</sub> through increased use of other transport modes such as public and non-motorised transport.
- 7.2.7 ULEZ, through the introduction of hybrid buses, can offer reductions in noise levels where compared to conventional petroleum-based engines.
- 7.2.8 ULEZ would have a positive impact on a variety of habitats and nature conservation sites in London, particularly woodland, grassland, heathland and wetland habitats.
- 7.2.9 ULEZ would have positive impacts upon cultural heritage features in London through reductions in NO<sub>x</sub> emissions, which can cause acid rain, and reductions in PM<sub>10</sub> which can cause soiling and discolouration of historic buildings.

### (b) Health and wellbeing

7.2.10 Improved air quality would make central London a more pleasant place to work, live and visit and would encourage personal health and wellbeing (e.g. increased use of public transport, walking and cycling as alternative modes of travel).



## (c) Population and equality

7.2.11 ULEZ would have a positive differential impact on school age children, older people and pregnant women due to a reduction in the number of sensitive receptors (e.g. residential properties, hospitals and schools) which are located in areas that experience exceedances of NO<sub>2</sub> AQO.

### (d) Economic

7.2.12 ULEZ would result in positive impacts on the economy arising from improved health and reduced impacts on the NHS. The improved health outcomes arising from the reduction in  $NO_X$ ,  $PM_{10}$  and  $PM_{2.5}$  under the ULEZ for the GLA area are estimated to have a total monetised benefit of £101m in 2020 and £32m in 2025.

## 7.3 Other impacts of ULEZ

7.3.1 This IIA acknowledges that there are likely to be some potentially negative consequences of ULEZ as follows.

### (a) Environment

- 7.3.2 An increase in demand for low and zero emission vehicles as a result of the ULEZ could have environmental impacts in terms of the material inputs required to manufacture them and the waste materials that would be produced from disposal of existing vehicle fleets.
- 7.3.3 Introduction of new low and zero emission bus, taxi and PHV fleets may have a negative impact on the landscape and urban realm as it would create demand for electric charging facilities and associated infrastructure across London. New infrastructure may also be required to support hydrogen vehicles, as these would need to be refuelled. Sensitive design of this infrastructure would be important in minimising any impact on London's landscape and streetscape.

### (b) Population and equality

- 7.3.4 ULEZ may have a differential impact on some equality groups, including:
  - female night workers in central London who currently drive to work by car and would be unable to afford a ULEZ compliant vehicle or pay the charge;
  - school children from low income families as a result of increased costs in school trips by coach;
  - disabled persons who might have difficulties finding an alternative mode of accessible transport to central London;
  - women and the LGBT community who fear for personal safety as a result of potential decrease in available taxis;
  - older taxi drivers who may choose to retire early rather than upgrade to a ULEZ compliant vehicle;
  - BAME PHV drivers where the PHV is registered to a SME; and
  - those groups that rely on charitable or voluntary services to access central London.



### (c) Economic

7.3.5 ULEZ would result in an economic cost felt by SMEs as well as the tourism sector due to the financial implications for coach operators. A number of mitigation measures have been identified that would help minimise these.

## 7.4 Mitigation and enhancement

- 7.4.1 TfL are already proactively seeking ways to mitigate potential impacts of ULEZ (in addition to those mitigation measures they have already committed to and that were embedded into the assessment of the ULEZ), these include:
  - investment in public transport, walking and cycling infrastructure including (but not limited to) increasing the coverage and frequency of night bus services, 24 hour London underground services on some lines and promoting a cycling revolution;
  - utilising existing CCZ infrastructure (e.g. cameras and signs);
  - promotion of the grant from the OLEV which provides a financial incentive for businesses with a UK address that need to replace vehicles to become compliant with ULEZ proposals; and
  - promotion of the Mayor's Air Quality Fund which provides for local boroughs to apply to become Cleaner Air Boroughs and use funding towards delivering innovative air quality improvement projects.
- 7.4.2 In addition, we recommend TfL take the following actions to ensure that any negative impacts identified in this report are fully understood and mitigated.

#### (a) Early and proactive engagement and research:

- 7.4.3 TfL should engage in early and proactive communication of the ULEZ. In particular, TfL should:
  - consult with the GLA, the Department for Environment, Food and Rural Affairs, the Environment Agency and waste management facility operators on waste management strategies to ensure any TfL policies encourage safe storage, reuse and disposal of hazardous vehicle components as already specified in EU regulations;
  - work closely with technology providers to ensure new infrastructure (e.g. charge point locations) meets London's needs and is in keeping with existing streetscape where possible;
  - work with the taxi industry to establish likely take up rates of a proposed financial incentive scheme for taxi replacement;
  - consult with charitable organisations to understand type and age of vehicle fleet and frequency of trips in CCZ;
  - work with SME business representatives in retail and wholesale sectors to identify potential measures to help mitigate impacts on them;
  - conduct stated preference research to understand likely mode shift response of different groups to a reduction in availability of taxis in central London;
  - promote environmental benefits of ULEZ and implementation of complementary policies (e.g. relating to improving London's air quality) to residents of Greater London and visitors;
  - undertake further work to fully assess and understand the implications for taxi fares as a result of any change in purchase price of taxis; and



 investigate further ways of working with the Government to provide financial assistance to operators of vehicles to replace non-compliant vehicles including support for the Mayor's call for a diesel scrappage scheme and investigation of its application to HGVs.

### (b) Recommendations for enhancement:

- 7.4.4 TfL should also consider the following as ways to enhance the benefits of ULEZ:
  - investigate the potential expansion of ULEZ / raising of ULEZ standards in the future (i.e. post 2025). This could result in further air quality benefits;
  - investigate and adopt recycling options for vehicle components from replaced existing fleets and encourage private vehicle owners to do the same;
  - for TfL vehicles, utilise existing facilities for disassembling batteries and recovering valuable materials such as cobalt and copper;
  - examine ways of providing financial assistance at a national level for small businesses to replace non-compliant vehicles with compliant vehicles; and
  - lobby for an extension of the existing OLEV grant should it expire. While guaranteed until 2020, OLEV has reserved the option to review the car grant value in 2017 or once 50,000 cars have been sold, whichever comes sooner.



## 8 References Greater London Authority (GLA). (2010). Mayor's Transport Strategy. May 2010. Accessed from: https://www.london.gov.uk/priorities/transport/publications/mayorstransport-strategy Greater London Authority (GLA). (2013). 2020 Vision: The Greatest City on Earth. Ambitions for London by Boris Johnson. London: Greater London Authority. Accessed from: https://www.london.gov.uk/sites/default/files/2020 vision web.pdf Greater London Authority (GLA). (2013). Mayor announces air quality game changer. Accessed on 9 October 2014 from: http://www.london.gov.uk/media/mayor-press-releases/2013/02/mayor-of-londonannounces-game-changer-for-air-guality-in-the Greater London Authority (GLA). (2014). Mayor backs national diesel car scrappage scheme as London ranked 9<sup>th</sup> best world city in new air quality study. 10 September 2014. Accessed on 16 October 2014 from: https://www.london.gov.uk/media/mayorpress-releases/2014/09/mayor-backs-national-diesel-car-scrappage-scheme-aslondon MVA consultancy in association with ERM and Future Inclusion. (2009). Draft Revised Mayor's Transport Strategy: Integrated Impact Assessment. Report for Transport for London, October 2009. Accessed from: http://www.london.gov.uk/sites/default/files/MTS%20IIA.pdf Transport for London (TfL). (2014). Health action plan. Transport for London (TfL). (2014). Transport Emissions Roadmap, cleaner transport for a cleaner London. September 2014. Accessed from: http://www.tfl.gov.uk/cdn/static/cms/documents/transport-emissions-roadmap.pdf Transport for London (TfL). (2014). Mayor's Air Quality Strategy. Accessed on 16 October 2014, from: http://www.cleanerairforlondon.org.uk/policy/mayors-air-qualitystrategy Transport for London (TfL). (2010). Cleaning the Air. The Mayor's Air Quality Strategy. Accessed on 16 October 2014, from: http://www.london.gov.uk/sites/default/files/MAQS%20Executive%20Summarv%20F INAL.pdf



#### 9 Acronyms AQO Air Quality Objective BAME Black, Asian, Minority Ethnic CAZ Central Area Zone CCZ **Congestion Charging Zone** $CO_2$ Carbon Dioxide EA **Environmental Assessment** EBIA Economic and Business Impact Assessment EqIA **Equalities Impact Assessment** EU **European Union** GLA Greater London Authority GLAA Greater London Administrative Area HGV Heavy Good Vehicle HIA Health Impact Assessment IIA Integrated Impact Assessment IRR Inner Ring Road LAEI London Atmospheric Emissions Inventory LGBT Lesbian, Gay, Bisexual and Transgender Light Goods Vehicle LGV LV Limit Value MAQS Mayor's Air Quality Strategy MTS Mayor's Transport Strategy NHS National Health Service Nitrogen Dioxide NO<sub>2</sub> Office for Low Emission Vehicles OLEV PHV **Private Hire Vehicle** ΡM **Particulate Matter** SEA Strategic Environmental Assessment SME Small to Medium Sized Enterprise TERM **Transport Emissions Roadmap** TfL Transport for London ULEZ Ultra Low Emission Zone



Appendix 1 – IIA framework for the ULEZ

Assessment	IIA Topic	IIA Objective	IIA Sub-Objective	Assessment Guiding Questions
Environmental	Air quality	To contribute to a reduction in air pollutant emissions and compliance with EU limit values (from Scoping Report).		<ul> <li>Reduce emissions to air?</li> <li>Reduce exposure to air pollution across London and address the disproportionate impacts felt by socio-economically disadvantaged communities?</li> <li>Contribute to effective traffic management to reduce local air pollutant emissions?</li> </ul>
	Noise	To reduce disturbance from general traffic noise (from Scoping Report).		<ul> <li>Reduce noise levels?</li> <li>Contribute to effective traffic management to reduce local noise emissions?</li> </ul>
	Climate Change	To reduce CO2 emissions and contribute to the mitigation of climate change (from Scoping Report).		<ul> <li>Promote the use of cleaner technologies and renewable energy?</li> <li>Promote smart travel options for all, promoting more sustainable modes of travel?</li> <li>Help develop more efficient and sustainable freight transportation?</li> <li>Encourage more efficient business and commercial supply patterns?</li> <li>Encourage uptake of green/cleaner fuels and energy sources?</li> </ul>
	Biodiversity including flora and fauna	To protect and enhance the natural environment, including biodiversity, flora and fauna (MTS IIA Objective F4).		<ul> <li>Protect and enhance local biodiversity?</li> <li>Protect local flora and fauna?</li> </ul>
	Cultural Heritage	To protect and enhance the historic, archaeological and socio-cultural environment (MTS IIA Objective F / F3).		<ul> <li>Protect designated and non-designated key historic, archaeological and cultural features or assets of value through inclusive design and management?</li> <li>Improve the use of the urban public realm by improving its attractiveness and access for all?</li> </ul>
	Water	To protect and enhance riverscapes and waterways through planning and operations (MTS IIA).		<ul> <li>Seek to minimise new development in areas prone to flood risk or mitigate the potential for such risk?</li> <li>Protect and enhance the character and use of London's riverscapes and waterways?</li> <li>Contribute to the sustainable use of waterways for passenger and freight transport?</li> </ul>
	Material Assets	To promote more sustainable resource use and waste management (MTS IIA Objective F1).		<ul> <li>Promote smart travel options for all, including reducing distance travelled and the need to travel, as well as promoting more sustainable modes of travel?</li> </ul>

Assessment	IIA Topic	IIA Objective	IIA Sub-Objective	Assessment Guiding Questions
	Townscape/Landscape	To protect and enhance the built environment and streetscape (MTS IIA Objective F2).		<ul> <li>Protect and enhance the built environment around key transport facilities, including removing barriers to use?</li> <li>Protect and enhance the character, integrity and liveability of key streetscapes, including removing barriers to use?</li> <li>Promote integrated, improved access for all within existing built environments and their landscapes through inclusive design and management?</li> <li>Protect and enhance valued/important built environment and streetscape settings through inclusive design and management?</li> <li>To promote active travel within streetscapes and surrounding environments?</li> </ul>
Health	Health and Wellbeing	To contribute to enhanced health and wellbeing for all within London	To address health inequalities and factors which negatively impact upon health and wellbeing	<ul> <li>Help to reduce health inequalities and key contributory factors to this?</li> <li>Support the physical and mental health and wellbeing of communities, particularly those disproportionately affected by inequality?</li> <li>Address factors which can negatively impact upon health and wellbeing, including:         <ul> <li>Reduce annoyance caused by transport noise (air, rail, underground and road traffic)?</li> <li>Reduce exposure to air pollution?</li> <li>Improve the quality of the travelling experience for all users and potential users of the London transport system?</li> <li>Reduce or mitigate community severance through sustainable transport planning?</li> </ul> </li> </ul>

Assessment	IIA Topic	IIA Objective	IIA Sub-Objective	Assessment Guiding Questions
			To promote enhanced health and wellbeing for all	<ul> <li>Help promote enhanced health and wellbeing through addressing key influences of health, including:         <ul> <li>Access to healthier and sustainable travel options including walking and cycling?</li> <li>Access to safe transport facilities and services?</li> <li>Access to employment and training?</li> <li>Access to local/community facilities?</li> <li>Access to local/community facilities?</li> <li>Access to local sport and cultural facilities?</li> <li>Access to local community facilities?</li> <li>Improve access for all, in particular, Deaf, disabled and older people, through the use of inclusive design to support sustainability?</li> </ul> </li> <li>Encourage the use of public transport by all sections of the community – including actions to promote access to</li> </ul>
			Improve air quality and the noise climate across London	<ul> <li>Reduce exposure to air pollution across London and address the disproportionate impacts felt by socio-economically disadvantaged communities?</li> <li>Contribute to effective traffic management to reduce local air pollutant emissions and noise levels?</li> <li>Encourage the use of more sustainable travel options and modes of transport such as public transport, walking and cycling and reduce car dependency and use, across all London's communities?</li> <li>Promote uptake of greener/clean technologies and renewable energy provision across all transport providers and private car users?</li> </ul>
Equalities	Population and Equality	To enhance equality and social inclusion	To ensure no protected and disadvantaged residents of London receive disproportionate or differential adverse impacts from traffic, emissions and noise as a result of the scheme.	<ul> <li>Reduce emissions to air and noise, experienced by protected and disadvantaged residents of London?</li> <li>Facilitate social inclusion through the removal of barriers?</li> <li>Minimise the adverse impact of freight transport on protected groups and disadvantaged communities</li> </ul>

Assessment	IIA Topic	IIA Objective	IIA Sub-Objective	Assessment Guiding Questions
			To give all users and potential users equal opportunity to access the London transport system and sustainable transport choices	<ul> <li>Encourage the use of public transport, walking and cycling by all sections of the community?</li> <li>Improve the accessibility of communities/areas across London, in particular, those experiencing exclusion?</li> <li>Enhance access for individuals to employment and training opportunities, in particular for those protected and disadvantaged?</li> <li>Take into account the different experiences of all users and potential users of London's transport system when planning, designing and delivering services?</li> <li>Reduce the barriers for Small and Medium Enterprises owned by those protected and disadvantaged groups?</li> <li>Increase accessibility to key services and facilities for all?</li> </ul>
Economic and Business	London's economic competitiveness	Provide an environment that is conducive to economic growth	Provide high quality environment will help to attract and retain internationally mobile businesses	<ul> <li>Provide an environment that will help to attract and retain internationally mobile businesses</li> <li>Control costs to doing business in central London</li> <li>Encourage the use of innovative ways to deliver goods and services in central London</li> <li>Attract employees from across the city</li> </ul>
	Small to Medium sized Enterprises (SMEs)	Support the growth and creation of SMEs	Encourage the development of new businesses in new and growth sectors	<ul> <li>Reduce the barriers for Small and Medium Enterprises</li> <li>Attract entrepreneurs through a high quality environment</li> </ul>